

SUNDAY DRAVERS, OR TOO FAST AND TOO FURIOUS?

ANNYANG SZEED, RIDEZ BEILINOUZ, AND TRAFFIC CONFLICTS OF E-SCOOTEZ SIMIZE

Author: Juan Francisco Arellano P.Eng. May 2019



Sunday Drivers, or Too Fast and Too Furious?

Analyzing Speed, Riding Behaviours, and Traffic Conflicts of E-Scooter Riders in Downtown San Jose, CA

A Planning Report

Presented to

The Faculty of the Department of

Urban and Regional Planning

San José State University

In Partial Fulfillment

Of the Requirements for the Degree

Master of Urban Planning

Ву

Juan Francisco Arellano P.Eng.

May 2019

ACKNOWLEDGMENTS

First and foremost. I would like to acknowledge the support and encouragement from my husband Adam; without him, I would have not been able to pursue a whole new career path in urban planning. I would like to thank my research supervisor, Dr. Kevin Fang, for his encouragement to pursue this research effort and for the opportunity to present my research findings at the 2019 Annual Transportation Research Board Meeting. Moreover, I would also like to thank my Planning Report Advisor, Rick Kos, for his meticulous and thoughtful mentoring and guidance throughout the composition of this report.

I would like to extend my personal gratitude to the members of my URBP 298B group (Tonya, Danielle, Carolyn, and Evan) for keeping me accountable through the writing process. Finally, I would like to thank my friends Clelia and Fran – I would not have been able to get through this program without their support.





Executive Summary	Ø		3.1.2 By Maximum Operational Technical Specifications 3.1.3 By the Numbers: Limiting E-Scooter Units	36 36
CHAPTER I. THE E-SCOOTERS ARE COMMON	21	3.1.4 3.2 3.3	Absolutely Not! Temporary E Scooter Bans E-Scooter Safety Regulation Considerations: Riding Behaviour Operational Characteristics in Similar Modes of Mobility and Planning	36 37 37
CHAPTER 2. THE SCOOTORNAL	27	3.4	MEASURING THE OBSERVATIONAL CHARACTERISTICS OF MICRO-MOBILITY DEVICES 3.4.1 VIDEO RECORDING 3.4.2 SITE DESERVATIONS GPS SUBVEYS AND SWART PHONE ADDUCATIONS	37 37 30
2.1 A BRIEF INTRODUCTION TO SHARED MICRO-MOBILITY 2.2 WHAT THE SCOOT IS AN E-SCOOTER? 2.3 THE E-SCOOTER OUTBREAK: E-SCOOTER CITIES AND OPERATORS	27 27 28	3.5 3.6	OPERATIONAL CHARACTERISTICS FOR MEASUREMENT: SPEED DATA ANALYSIS REDUCTION: AGE, GENDER, AND TRAVEL BEHAVIOURS (CONFLICTS)	39 39 40
2.4 THE E-SCOOTER SHARE SERVICE MODEL OF OPERATION 2.5 NOT ALL E-SCOOTERS ARE BUILT THE SAME	28 28	3.7	Why The Scoot Should Cities Care	41
2.6 Scoutering Along: Signing-Up, Starting, Traveling, and Ending Your Ride 2.6.1 Signing-Up 2.6.2 Structure & Ride	30 30 20	(]]]? 4.1	PTER 4. METTOOS Speed	45 45
2.6.4 ENDING THE RIDE	30 30 32	4.2	RIDING BEHAVIOURS 4.2.1 HELMET USE 4.2.2 TEANEL STRUGHT FORWARD OR SIDE TO SIDE	47 47 47
2.7 The Scootorial Findings	32		4.2.3 GROUP RIDING/MULTIPLE RIDERS 4.2.4 TRAFFIC CONFLICTS	47 49
CILIPITER 3. WHITE THE SCOOT IS GOING ONE 3.1 Regination Departes: What are Cities Doing to Reginate E-Scooter Surge	35 25	4.3	4.2.5 RIDER DISTRACTIONS ENVIRONMENTAL CONSIDERATIONS: WET, DRY WEATHER, AND TEMPERATURE	49 49 40
3.1.1 FACILITY DEBATES: WHERE TO SCOOT? STREETS, SIDEWALKS, OR MIXED-USE PATHS?	35	4.4	ANALYSIS REDUCTION METRICS: AGE, APPARENT GENDER METHODOLOGY SUMMARY	



CONTENTS

(mmassa & Mas an Briss man Missonasa	6.5 TRAFFIC CONFLICTS ON SIDEWALKS AND MIXED-USE PATHS	
Gulil-Mers 25 Mon 30 March Mons Charlistica	6.6 Increasing the Risk of Injury? E-Scooters and Vehicle Collisi	ONS
5.1 KIDERS IRAVEL FASTER ON STREETS, SLOWER ON SIDEWALKS, BETWEEN 9.0-11.1 MPH	53 6.7 VOTING WITH AN UNPROTECTED HEAD: E-SCOOTERS AND HELMET US	E
5.2 WIDE RANGE OF RIDER SPEED BETWEEN 4 AND 16 MPH	53 6.8 E-Sconter Riders and Distructions are not so Much de a Pro	RIEM
5.3 MALES RIDE FASTER, VARY SPEED LESS BY FACILITY	$55 \qquad \qquad 60 E = 50000000 \text{ And } 00000000000000000000000000000000000$	
5.4 No Difference by E-Scouter Share Operator	56 6 10 Device the American America Converting For District Converting	
5.5 OLDER RIDERS TRAVEL FASTER THAN THEIR YOUNGER COUNTERPARTS	58 C 11 Due Man Theorete Man and Conventions	
5.6 E-Scooter Riders Travel Slower than Cyclists	58 O.II BUT WAIT, IHERE'S MORE!	
	60 6 11 2 TUDNING STORMUS	
5.8 MADE EVELISTS WERD HELMETS THAN E-SCARTED DIALOS	$\frac{60}{60} \qquad \qquad 612 \qquad \qquad 610 \qquad $	
5.0 MORE CICLISIS MEAR HELMEIS HAAN L'SCOULER RIDERS	ED COLLE IMPLICATIONS JUMMART	
5.5 A NUTREABLE AMOUNT OF GROUP RIVING		
S.IU MINIMAL KIDER UISTRACTIONS: SOME WEAR MEADPHONES, VERY FEW USE CELLPHONES	02 COMPETER 7 Des Prinning same Contra Contra	
S.II IRAFFIC CONFLICTS: SLOWING DOWN VS. SWERVING OUT OF THE WAY		11
5.12 Most E-Scooter Riders Travel in a Straight Line	64 7.1 ALLOW SIDEWALK SCOOTERING WHERE AND WHEN IT MAKES SENSE	USING
5.13 Some Riders Fall While Others Collide	64 7.2 TO AVOID COLLISIONS AND SIDEWALK KIDING, SEPARATE MICRO-MOB	ILITY
5.14 The Results: E-Scooter Riders in San Jose	64 7.3 Building Slow, Medium, and Fast Lanes to Accommodate All Ti	TPES C
	7.4 INCREASE THE WIOTH OF MICRO-MOBILITY INFRASTRUCTURE TO PRO	DMOTE
	7.5 AGE REGULATION: RE-THINKING MANDATORY HELMET LAWS	
GALAPITER OF CONCLUSIONS AND INTERIOUS	051 7.6 FROM THE BICYCLE LANE TO THE "GREEN" LANE	
6.1 Not so Fast and Furious, Speeds are Similar to Other Modes		
6.2 CREATING NEW CONFLICTS: SLOW SCOOTERS AND FAST CARS	70	
6.3 IS AGE TRULY THE BEST WAY TO REGULATE E-SCOOTERS?	70	
6.4 About Those Pesky Sidewalk Riders on Santa Clara St.	70	

POSTED SPEED LIMITS DEVICES FROM VEHICLES OF RIDERS E E-Scooter Riding





CHAPTER 3. UNITATIONS AND FUTURE RESEARCH	3
8.1 SPEED RESULTS 8.2 PIOLNG REHAVIOURS	83 83
8.3 SCOOTING FORWARD: FUTURE RESEARCH	83 84
8.3.2 A GENOER IMBALANCE	84
ENDNORTES	Ĩ
BIRUMERINDITT	94





TABLE 1. SPEED OF PEDESTRIANS, CYCLISTS, AND OTHER MICRO-MOBILITY DEVICES TABLE 2. DIFFERENCES IN SPEED BY FACILITY AND GENDER 38 57



FIGURES

FIGURE 1. E-SCOOTER OBSTRUCTING THE ENTRYWAY TO ORIGINAL GRAVITY IN DOWNTOWN SAN JOSE, CA.	22	FIGURE 23. PER
FIGURE 2. E-SCOOTER IN A TRASH RECEPTICLE IN SAN JOSE, CA.	23	FIGURE 24. PER
FIGURE 3. E-SCOOTER IN THE GUADALUPE RIVER IN SAN JOSE, CA.	23	FIGURE 25. PERC
Figure 4. Wind e-scooters knocked over in San Jose, CA.	23	
FIGURE 5. BIRD AND LIME E-SCOOTER OPERATORS IN SAN JOSE, CA.	29	
FIGURE 6. SCOOT'S LOCK-TO LOCK IN SAN FRANCISCO, CA.	30	
FIGURE 7. SKIP SCOOTER IN SAN JOSE, CA.	30	
FIGURE 8. LIME E-SCOOTER QR CODE.	30	
FIGURE 9. U.S. CITIES FEATURING E-SCOOTER PROGRAMS.	31	
FIGURE 10. STEP 1-SELECT YOUR E-SCOOTER.	32	
FIGURE 11. STEP 2-READ THE RULES.	32	
FIGURE 12. STEP 3-LOCK YOUR RIDE.	32	
FIGURE 13. SIDEWALK AND STREET OBSERVATION ZONE LENGTHS ALONG SANTA CLARA ST. IN SAN JOSE, CA.	46	
FIGURE 14. OBSERVATION ZONE LENGTH ALONG THE MIXED-USE PATH AT SAN JOSE STATE UNIVERSITY.	48	
FIGURE 15. SPEED RESULTS ACROSS ALL THREE TRANSPORTATION FACILITIES.	54	
FIGURE 16. DISTRIBUTION OF E-SCOOTER RIDER SPEEDS ON DIFFERENT TYPES OF TRANSPORTATION FACILITIES.	55	
FIGURE 17. DISTRIBUTION OF E-SCOOTER SPEEDS BY OPERATOR.	56	
FIGURE 18. DISTRIBUTION OF E-SCOOTER RIDER SPEEDS BY AGE IN YEARS.	<i>58</i>	
FIGURE 19. SPEED RESULTS FOR E-SCOOTERS AND CYCLISTS.	<i>59</i>	
FIGURE 20. DISTRIBUTION OF E-SCOOTER RIDER SPEEDS ORDERED BY TEMPERATURE RANGES.	60	
FIGURE 21. PERCENTAGE OF E-SCOOTER RIDERS AND CYCLISTS THAT WEAR HELMETS.	61	
FIGURE 22. PERCENTAGE OF E-SCOOTER RIDERS TRAVELING SOLO OR IN GROUPS.	62	

RCENTAGE OF MULTIPLE RIDERS SHARING ONE E-SCOOTER. RCENTAGE OF E-SCOOTER RIDERS USING CELL PHONES. RCENTAGE OF RIDERS USING HEADPHONES. EXECUTIVE SUMMARY

In recent months, mobility companies have been rapidly introducing shared electric scooter (e-scooter) programs in cities across the United States. While there appears to be strong interest, they have attracted a wide range of opinions. Some view e-scooters favourably, specifically pertaining to their benefit of providing alternative forms of mobility as opposed to driving, and the potential to address first-mile and last-mile access.¹ Others view them as nuisances "obstructing entryways" or creating "clutter".²

Shared e-scooter systems are still in the early stages of deployment and many governments are just beginning the process of debating how scooters should be regulated.³ This debate over whether individuals view e-scooters as compatible or incompatible with other road users and whether they support or do not support specific policies is likely formed by how they see the attributes of e-scooters versus that of similar modes of micro-mobility. However, there is a lack of data on how e-scooters riders actually behave, and how that compares to other road users. As a result, this study sought

to observe e-scooters on sidewalks. streets, and mixed-use paths in San Jose with a goal of revealing how operational characteristics, riding behaviours, and traffic conflicts could inform those debates.

RESEARCH QUESTION

This research effort sought to answer the following question:

What are the "operational characteristics" exhibited by e-scooter share users using streets, sidewalks, and mixed-use paths in downtown San Jose. California and what are the potential implications of the findings on the future development of e-scooter regulation and micro-mobility (bicycle) infrastructure design?

To address this gap in research, the operational characteristics of e-scooter users were investigated including speed, and other qualitative measures of behaviour that might influence safety such as helmet use, riding style, group travel, traffic conflicts, and rider distraction

METHODS

In total, 330 e-scooter riders were

observed in downtown San Jose during a mix of both dry and wet weather conditions on streets (n=110), sidewalks (n=110), and mixed-use paths (n=110)between October 2018 and February 2019. Only Lime's Lime-S, and Bird scooters were observed as part of this study. Additionally, 110 observations of cyclists were observed on the street only.

RESULTS

Concerning average speed:

- riders traveled between 9 to 11 mph (p<0.01) with a different speed per facility
- male riders traveled faster than females and varied less by facility (p<0.01)
- operator speeds were similar at 10 (p=0.69) mph
- older adult riders (10.5 mph) traveled faster than younger riders (9.4 mph) (p < 0.01)
- e-scooter riders traveled slower on streets (11.1 mph) than bicyclists (12.2 (10.0) (hdm
- riders traveled faster in colder temperatures than warmer temperatures (p=0.04)

behaviours:

- cvclists

- cellphone
- using headphones

In terms of traffic conflicts (n=134). 71 percent of riders that encountered conflicts slowed down while 29 percent swerved out of the way. Furthermore, 70 percent of riders who encountered conflicts on facilities with mixed pedestrian traffic slowed down for pedestrians (a positive finding). Of note, the few collisions observed (n=3), were as a result of a rider colliding with a curb or due to a rider colliding with a parked car's side view mirror. This finding was similar to that of skateboarder collisions.

IMPLICATIONS

Based on the observation results, this study looked at the potential implications as it pertained to regulatory debates over e-scooters and the design of urban infrastructure and found: • riders are not traveling as fast as

Furthermore, concerning riding

• 2 percent of e-scooter riders wore helmets compared to 55 percent of

• 97 percent traveled in a straight line as opposed to traveling in a less predictable side to side motion • 16 percent of riders traveled in groups • 3 percent of riders traveled with two riders on one scooter

• only one person was seen using a

• 16 percent of riders were observed

what some people perceive, and are comparable to other modes of micromobility (where some are permitted to operate on sidewalks)

- new conflicts are created as a result of pitting slow scooters against fast moving vehicles
- age may not be the best way to regulate e-scooter share due to the inconsistency of riding behaviours among different age groups regarding speed
- sidewalk riders are yielding to pedestrians on sidewalks
- placing e-scooters riders on street could potentially be increasing their risk of collisions
- people are not distracted by cell phone use, but could be potentially distracted from headphones, and be violating state laws
- with helmet compliance being low among all riders, existing laws need to be revisited as they are not working towards encouraging more helmet use
- there was evidence of talking while observing group riding, suggesting a potential for distractions
- e-scooter share is not the only observed emerging mode of mobility; cities might have to rethink naming conventions for existing bicycle infrastructure

RECOMMENDATIONS

Based on a review of e-scooter riding results of speed, riding behaviours,

and traffic conflicts, along with an understanding of the implications, this study made seven recommendations for the regulation of e-scooters along with design of urban micro-mobility infrastructure:

- allow sidewalk riding where it makes sense by using posted speed limits
- to avoid sidewalk riding, build safer infrastructure on streets
- to avoid e-scooter collisions, separate vehicle traffic
- building multi-speed lanes to accommodate all types of riders
- increase the width of existing micromobility infrastructure to promote more e-scooter use
- reconsider age restrictions concerning helmet laws
- rename the bicycle lane to the green lane

Research Limitations and Future Research

Additionally, this research effort discussed both research limitations and future research considerations. Concerning the speed results, observed riders on the mixed-use path (paseo off of 7th Street) were younger in age due to the proximity of the street facility near San Jose State University. This may have resulted in over sampling of similar age ranges. Also, there might have been a potential for human error due to the reaction time required to start and stop the stopwatch as riders crossed the observation zone. As it pertains to determining the age of

e-scooter riders, there is some subjectivity over the classification of observed riders by age due to the observer determining these age ranges. As a result, there might be a larger proportion of riders represented in the adult category (25 to 50 years of age) that could have fallen either into the younger (adolescent) category of less than years of age, or the older adult category of individuals over the age of 50.

Moving forward, there is still much to learn about shared e-scooter programs and e-scooters in general. In terms of operational characteristics, data on the ability of e-scooters to brake and maneuver could be a particularly useful topic for future research. For example, what stopping distance is required for an e-scooter rider on a sidewalk to react to a sudden movement by a nearby pedestrian? Could an e-scooter rider navigate facilities with certain design specifications? Moreover, regarding safety, do e-scooter riders refuse to wear helmets for the same reason as cyclists?





In recent months, mobility companies have been rapidly introducing shared e-scooter programs in cities across the United States. The arrival of shared e-scooters continues the trend toward shared mobility (such as bike share, ride share, and car share) and "micromobility"— low-speed transportation for individual travelers on short trips.⁴ Populus, a U.S. transportation analytics firm, finds strong interest in shared e-scooters from the public.⁵ Populus estimates that in less than one year since their introduction, 3.6 percent of adults in U.S. cities where shared e-scooters are available have used the service. This estimate points to a much more rapid adoption rate than seen in other shared micro-mobility services.⁶

THE E-SCOOTERS ARE COMMON

CHAPTER 1 THE ESCOOTERS ANRE COMMING

While there appears to be strong interest with the arrival of e-scooters, they have attracted a wide range of opinions. Some have responded to e-scooters favourably. specifically pertaining to their benefit of providing alternative forms of mobility as opposed to driving, and the potential to address first-mile and last-mile access.⁷ Moreover, some have pointed to the ability of these power-assisted devices to assist riders in overcoming physical obstacles.⁸ The Populus study also finds that scooters are popular among the public at large with approximately 70 percent of survey respondents in 11 major U.S. cities holding favourable views of e-scooters.⁹ On the other hand, there are those that have reacted less favourably, viewing e-scooters as nuisances "obstructing entryways" or

creating "clutter".¹⁰ Beverly Hills, California Mayor Julian Gold, claimed scooters "put everybody at risk, and they put your kids at risk, and there's no responsibility for it, at all".¹¹ Moreover, there is an issue concerning the public perception over the safety of these devices with injuries associated with e-scooter riding.¹² Recently, San Jose State University's Chief Financial Officer, Charlie Faas, was quoted stating that e-scooters riders pose a safety risk due to the fact people are riding on sidewalks and not wearing helmets.¹³ Figures 1, 2, 3, and 4 depict some of the issues associated with e-scooter share such parking, litter, and blocking the sidewalk.

One of the more contentious points about e-scooter operation pertains to



the compatibility of scooters with other road users, particularly pedestrians on sidewalks.¹⁴ Some jurisdictions view e-scooters as strictly incompatible with pedestrians. For example, California state law prohibits e-scooters outright on sidewalks.¹⁵ Similarly, other jurisdictions implicitly view e-scooters as incompatible with other road users. For example, the City of Denver, Colorado defines e-scooters as "toy vehicles" and mandates they be ridden on sidewalks.¹⁶ E-scooters are also prohibited on bicycle paths in Denver. Other cities such as Milwaukee and San Francisco have outright banned e-scooters altogether. at least on a temporary basis.¹⁷ As of August 30, 2018, San Francisco has lifted its e-scooter ban and instituted a pilot program.¹⁸



THE E-SCOOTERS ARE COMMON

the early stages of deployment and many governments are just beginning the process of debating how e-scooters should be regulated.¹⁹ Since e-scooter share has not been around for very long there is also a lack of literature on how best to regulate e-scooters, especially when there is a lack of understanding on how riders behave; however, there are methodologies for similar modes of mobility that may be used to establish comparisons for developing such a methodology. What literature is available, such as that from the National Association of City Transportation Officials (NACTO), codifies the day to day operation of "shared active transportation" devices from existing regulations based on rightof-way regulation, zoning regulation, small-vehicle regulation, and existing contracts with operators.²⁰ While NACTO's guide strives to inform cities on their

choices of regulating e-scooter share, the report relies on using a city's existing set of regulations that are not necessarily reflective of the operating characteristics of e-scooter share, nor rider behaviour.²¹

The argument over whether individuals view e-scooters as compatible or incompatible with other road users and whether they support or do not support specific policies is likely formed by how they see the attributes of e-scooters versus that of different modes. The maximum capabilities of most e-scooters are published in technical specification documents. However, there is a lack of data on how e-scooter riders actually behave, and how that compares to other road users.

To address this gap in research, this research effort investigated the physical



FIGURE 2. E-SCOOTER IN A TRASH RECEPTICLE IN SAN JOSE, CA. PHOTOGRAPHY BY THE AUTHOR.



FIGURE 3. E-SCOOTER IN THE GUADALUPE RIVER IN SAN JOSE CA. PHOTOGRAPHY BY AUTHOR.



GURE 4. WIND E-SCOOTERS KNOCKED OVER IN SAN JOSE CA. PHOTOGRAPHY BY THE AUTHOR.

"operational characteristics" of e-scooter users. The operational characteristics measured here include speed and other qualitative measures of behaviour that might influence safety such as helmet use, riding style, group travel, traffic conflicts, and rider distraction. Specifically, this report seeks to answer the following question:

What are the "operational characteristics" exhibited by e-scooter share users using streets, sidewalks and mixed-use paths in downtown San Jose, California and what are the potential implications of the findings on the future development of e-scooter regulation and micro-mobility (bicycle) infrastructure design?

Of note, this will be the first research effort the design of transportation facilities of its kind to investigate the operational characteristics and riding behaviours of e-scooter riders. The data measured to answer those questions came from downtown San Jose, California, where at least two companies have operated shared dockless e-scooter systems since early 2018. The data in this report reflects observations of e-scooter riders on three different kinds of transportation facilities: streets, sidewalks, and mixed/shareduse paths in both dry and wet weather conditions. Also examined are shared e-scooter systems currently utilizing motorized versions of "kick scooters," which are comprised of a long narrow platform that riders stand on, rolling on

two wheels, with a vertical beam at the front rising to handlebars.²² Note, only two e-scooter operators were included as part of this study, Lime (and their Lime-S e-scooter) and Bird.

Chapter 2 explores the arrival and development of micro-mobility and e-scooter share. Chapter 3, a literature review of e-scooter share, covers two important topics pertinent to this research effort: the regulatory debates over e-scooters in cities, and the measurement of operational characteristics and riding behaviours in similar modes of micromobility. Importantly, the reviewed literature suggests that the recording of operational characteristics may prove useful in the urban planning profession regarding and regulations for alternate modes of mobility.²³ Chapter 4 describes the project methodology, followed by a discussion of the results in Chapter 5. Chapter 6 covers the potential implications of the findings as they pertain to urban planning policy for regulating e-scooter share, e-scooter design impacts, and micromobility infrastructure design. Chapter 7 summarizes recommendations, and Chapter 8 covers research limitations and future research considerations.





THE SCOOTORIAL

CHAPTER 2 THE SCOOTORIAL

2.1 A BRIEF INTRODUCTION TO SHARED MICRO-MOBILITY

The introduction of shared active transportation vehicles has proven popular in U.S. cities as people discover alternative ways to travel to their destinations.²⁴ The National Association of City Transportation Officials (NACTO) defines shared active mobility devices as "a network of small vehicles" for rent used for travel over short distances.²⁵ Shared active transportation devices can also be referred to as "micro-mobility" devices such as "pedal or electrically powered bicycles, scooters, and mopeds."²⁶

The emergence of shared active transportation mobility devices in the U.S. commenced in 2009 with the introduction

of Capital Bikeshare in Washington, D.C.²⁷ Initial variants of this pioneering service featured systems that required users to use docking stations to secure their rental bicycles.²⁸ The evolution of bike share, "dockless bicycle share", can be traced back to 2011 with the introduction of Social Bicycles (SoBi) in Buffalo, NY. SoBi stood apart from existing bicycle sharing systems at the time as SoBi did not require a dock and therefore was a dockless-based system.³⁰

Micro-mobility evolved during 2016 and 2017 as new entrants Lime and Spin introduced their concepts of dockless devices.³¹ Unlike the existing dockless bicycle sharing systems at the time by Motivate (purchased by Lyft in June of 2018) and SoBi (now Jump, and

purchased by Uber in April of 2018), newer companies are all privately financed through venture capital.³² Bicycles, and more recently e-bikes, are operated on public rights-of-way.³³ NACTO notes that these initial entrants entered the market without the necessary permitting as required by many cities today.³⁴ One can attribute the birth of the e-scooter share phenomena to Bird in Santa Monica, CA. Bird launched its dockless shared e-scooter system in 2017. Similar companies began to spread across the U.S. including Spin, Bird, and Lime.³⁵

2.2 WHAT THE SCOOT IS AN E-SCOOTER?

An e-scooter is a motorized standing kick scooter with two wheels and handlebars

that features a long flat board that people stand on while they ride.³⁶ The first set of e-scooters distributed in San Jose, from both Lime and Bird, featured a manual handbrake on the left-hand side to manually brake the e-scooter. Additionally, • riders controlled the throttle by using an electronic thumb pad to accelerate and decelerate. In the next e-scooter wave in San Jose, both Lime and Bird opted to modify their e-scooter designs and move towards a custom designed Segway version.³⁷ The most noticeable difference is the additional battery packs to boost the operating performance of e-scooters in order to travel longer distances (35 miles).³⁸ Additionally, new e-scooters no longer have a manual brake lever on the handlebars; instead, there are two brakes on both wheels which are electronic and are controlled by a thumbpad similar to the accelerator. In order to apply the brake, riders have to press down on the brake thumbpad. See Figure 5 for examples of e-scooter operators in San Jose, CA.

2.3 THE E-SCOOTER OUTBREAK: E-Scooter Cities and Operators

As of 2018, the following U.S. cities feature e-scooter share programs:

- Austin, TX
- Boston, MA
- Charlotte, NC
- Chicago, IL
- Dallas, TX
- Denver, CO

- Los Angeles, CA
- Miami, FL
- Oakland, CA
- Palo Alto, CA
- Portland, OR
- Nashville, TN
- San Diego, CA
- San Francisco, CA
- San Jose, CA
- Santa Monica, CA
- Washington, D.C.

The U.S. market features a variety of e-scooter share companies including: Bird, Lime, Lyft (Motivate), Uber (Jump), Scoot, Skip, Spin, and Wind. Note, while Palo Alto features an e-scooter share program ordinance, it has yet to grant permits to operators.

2.4 THE E-SCOOTER SHARE SERVICE MODEL OF OPERATION

The current e-scooter share market provides an on-demand e-scooter service. Unlike previous bike share programs, monthly subscriptions are not available.42 Typically, operators offer their services starting with a flat fee of one dollar before charging a 15 cent rate for every minute after that.43 Bird deviates from this model and currently offers a promotion where riders can pre-book a delivery of an-e-scooter in the morning for commuting to work.44 Additionally, certain cities like Austin, Denver, San Francisco, Los Angeles, and Palo Alto impose affordability requirements on

e-scooter share operators to ensure equal spatial coverage across cities. For example, Austin requires that e-scooter share operators provide a non-mobile option to pay for services for anyone that is at or below the federal income poverty level.⁴⁵ Denver, on the other hand, is less specific about its equity requirements and instead leaves discount programs up to the operator to propose for those without smartphones and "unbanked" users.46 The California cities of Los Angeles, Palo Alto, and San Francisco provide a hybrid of the two earlier mentioned systems by stipulating a non-mobile (non-credit card) option for those below the poverty level, but also dictate that service will provide unlimited trips under 30 minutes while waiving any fees or security deposits.⁴⁷ In San Francisco. Lime proposed to provide student discounts by allowing students to register through their academic institution ".edu" email address.48

2.5 NOT ALL E-SCOOTERS ARE BUILT THE SAME

While most e-scooter share units appear to be similar, they do possess unique characteristics. In the City of San Francisco, CA, both Scoot and Skip's e-scooters feature a bike lock that is attached to the e-scooter itself.⁴⁹ The idea behind this addition is that users can lock e-scooters to surrounding street furniture, thus confining these devices to areas outside of the pedestrian travel path.⁵⁰ A key difference between the



two is that Skip's lock-to technology is a retractable steel wire whereas Scoot's lock-to technology is a standard bicycle lock with a combination.⁵¹ Furthermore, Skip touts their e-scooter as being "better built" with a custom design that features a wider standing board, dual suspension, adjustable handlebar height, and tail lights to tell those around the rider when they are slowing down.⁵² Additionally, when comparing Lime-S e-scooters to Bird, Lime-S e-scooters possess an electronic speedometer, while Bird e-scooters do not.

2.6 Scootering Along: Signing-UP. STARTING, TRAVELING, AND ENDING YOUR RIDE

The section discusses the e-scooter riding experience as it pertains to the sign-up process, travel experience, and



FIGURE 6. SCOOT'S LOCK TO LOCK IN SAN FRANCISCO, CA. PHOTOGRAPHY BY THE AUTHOR.

ending the ride.

2.6.1 SIGNING-UP

The sign-up process for an e-scooter starts by users downloading the operator's respective mobile phone application (see Figures 10, 11, and 12).53 Only Lime and Bird were included in this discussion as they are currently operating in the City of San Jose, CA. Both Bird and Lime require riders to provide identification (such as a driver's license) that users must scan with their smartphone's camera as part of the sign-up process.⁵⁴ The process also forces riders to swipe through rules and regulations regarding appropriate riding facilities, parking, and helmet use in the local jurisdiction before beginning to ride.⁵⁵ Once the legal agreements and regulation reminders have been reviewed and accepted, riders are free to start their ride. Once a rider has signed up for an



FIGURE T. SKIP SCOOTER IN SAN JOSE, CA. PHOTOGRAPHY BY THE AUTHOR.

30

e-scooter share mobile phone application, finding e-scooters is merely a matter of launching the application and searching the built-in map function to find the nearest e-scooter.56

2.6.2 STARTING A RIDE

Starting an e-scooter ride requires riders to scan a QR code which is typically located near the handlebars.⁵⁷ Once the code is scanned, the rider is required to wait a few seconds prior to the e-scooter unlocking. A jingle will sound to notify riders that they can now start their e-scooter ride (see Figure 8).58

2.6.3 RIDING EXPERIENCE

Depending on the type of surface of the transportation facility, e-scooters will provide a different riding experience. On smoother types of surfaces, a rider can expect a quiet and smooth enjoyable



FIGURE 8. LIME E.SCOOTER QR CODE. PHOTOGRAPHY BY THE AUTHOR.



Palo Alto, CA

1000 Miles



ride.⁵⁹ When moving onto a road however, are required to open the mobile phonethe experience can be a different matter. Currently, certain types of e-scooters from Segway do not feature much in the way of shock absorption. While riding on a rough street or sidewalk, one feels every single bump along the way. In wet weather conditions, riding conditions can vary widely as wet surfaces can prove to be slippery.

2.6.4 ENDING THE RIDE

Upon reaching a destination, riders are required to first park their e-scooters in a location that does not obstruct the sidewalk.⁶⁰ In San Jose, there are no specifically designated parking areas, unlike other cities where they provide painted boxes to park them. E-scooters also feature a kickstand that riders can engage to let the scooter right itself while parked. Once the vehicle is parked, riders



based application to end their ride.⁶¹ Riders tap their smartphone screen to end their ride and are provided with a summary of their trip length and travel time.⁶² Riders are also prompted to rate their ride experience from one to five stars. Lower starred rides prompt users to input feedback on how to improve their experience for the next time.⁶³ Additionally, riders are also required to provide a photo of where their e-scooter is parked for two reasons: to show the next rider where the scooter is located in case they cannot find it, and a form of self-policing to ensure scooters are parked in an area that is not obstructing the right-of-way.64

2.7 THE SCOOTORIAL FINDINGS

This chapter served as an introduction to the emergence of shared micro-

Do not ride on sidewalks other than as permitted by law. Use bike lanes when available and permitted by law.



mobility in the U.S. The "scootorial" started by looking at early shared micromobility systems such as bike-share before delving into e-scooter share. The chapter then discussed the availability of e-scooter share across the U.S. along with the operators of the service. The chapter concluded with a summary of the e-scooter share ride service experience from start to finish.

While this chapter covered the basics of e-scooter share in the U.S., there still remains many unanswered questions over regulatory debates of e-scooter share. Chapter 3 will not only cover what cities are doing to regulate e-scooter share, but also the methods most commonly used to regulate similar modes of micro-mobility such as bicycles, e-bikes, skateboards, and Segways.

End your trip by pressing the "Lock" button in the app.







A review of the most recent literature was behaviour data is currently non-existent. facilities, including sidewalks, pedestrian conducted to identify what governments paths, bicycle paths or lanes, and 3.1 REGULATION DEBATES: WHAT streets. Regulations in some areas limit (both municipal and state) in the United States are currently doing to regulate e-scooters to certain types of facilities and ARE CITIES DOING TO REGULATE not others. Existing policy or frameworks e-scooter share, and the relevance of E-Scooter Share are temporary, and what is available in measuring operational characteristics from a planning perspective by looking at A review of the most recent literature terms of literature is limited. similar modes of micro-mobility. Different regarding how cities are adapting to the For example, the state of California, e-scooter influx reveals a mixed bag of methodologies were then reviewed for as of mid-2018 prohibits the use of measuring operational characteristics, approaches ranging from imposing facilit riding behaviours, and analysis reduction restrictions, speed limitations, restricting e-scooters on sidewalks. During the 2018 strategies. Importantly, the reviewed the number of operators and scooter units legislative session, California Assembly literature points towards the notion that Bill 2989 was introduced with the goal and, in the most extreme cases, outright the recording of operational characteristics bans. of clarifying where and how e-scooters can operate. The first version of the bill may prove useful in the urban planning 3.1.1 FACILITY DEBATES: WHERE TO SCOOT? proposed permitting sidewalk operation.66 profession regarding the design of STREETS, SIDEWALKS, OR MIXED-USE PATHS? transportation facilities and regulations However, this was removed from later concerning alternate modes of mobility.⁶ In cities where e-scooter share currently versions of the bill over concerns from Furthermore, this addresses a gap in the operates, users can be seen operating some groups that the legislation would literature where e-scooter share travel result in e-scooters traveling 25 miles them on various types of transportation

MILET THE SCOOT IS GOILD ON?

CHAPTER 3 WWWWWWEFF FILLE SCOOLT US GOULANG OUNT?

per hour (mph) on sidewalks (the bill defined e-scooters as devices that could have up to a maximum speed of 25 mph).⁶⁷ This statewide approach to regulation, however incorporated clauses for cities to implement their own operation restrictions.⁶⁸ At the municipal level within California, cities such as Santa Monica, Los Angeles, and San Francisco have sought to restrict e-scooter share operation to streets and prohibit sidewalk operation as part of temporary pilot programs.⁶⁹ Amongst these three California cities, Santa Monica had the most restrictive legislation, confining all forms of e-scooter riding to streets only.⁷⁰ On the other hand, some cities implicitly view e-scooters as incompatible with other road users. For example, the City of Denver, Colorado defines e-scooters as "toy vehicles" and mandate they be ridden on sidewalks.⁷¹ Additionally, Denver also prohibits the operation of e-scooter share on bicycle paths.⁷² In the City of San Jose, e-scooters are banned from operating on sidewalks, while San Jose State University has completely banned the operation of all e-scooters on campus effective March 2019.73

3.1.2 By Maximum Operational Technical **SPECIFICATIONS**

In the absence of actual operational characteristic data, debates over the regulation of a device often falls upon the maximum capabilities of that device, as was the case in California over A.B.

2989. At the statewide level in California. e-scooters are limited to streets with a posted 25 mph per hour speed limit; with A.B. 2989, this speed was increased to 35 mph.⁷⁴ A brief glance at the technical specification documents for e-scooters reveal that they have maximum operating speeds between 15 and 18 mph.⁷⁵ Shared e-scooter operators typically limit the top operating speed of their scooters. For example, operators such as Lime and Bird limit the top operating speed to 15 mph.⁷⁶ In the City of Los Angeles, unlike the current state-level regulation. e-scooters are capped at a maximum operating speed of 15 mph as part of their pilot program.⁷⁷ Moreover, the City of San Jose also utilizes speed restrictions. limiting e-scooters to 12 mph.⁷⁸ As it pertains to speed, the State of California and Los Angeles have adopted two different approaches. California's approach regulates the facility e-scooters can operate on by the use of street speed limits, versus Los Angeles's approach of capping the maximum operational speed of e-scooters.

While technical specification documents are informational, basing regulations on them is not necessarily consistent with how other modes of transportation are regulated. For example, automobiles are generally not prohibited in residential neighbourhoods because they can reach speeds more than 100 miles per hour. Rather, cities apply behavioural

regulations in the form of speed limits. On the flip side, regulations on electric bicycles, including model regulations recommended by the League of American Bicyclists, which have been adopted in 20 states, regulate devices at least in part on the maximum capabilities of different e-bike models.79

3.1.3 BY THE NUMBERS: LIMITING E-Scooter UNITS

Another common method for regulating e-scooters has been limiting the number of units and operators available as part of daily operations. For example, the City of San Francisco limited its pilot program to two operators. Skip and Scoot, with 625 scooters each.⁸⁰ The City of Santa Monica allowed for more operators (Bird, Lime, Lyft, and Jump [Uber]) with 750 scooters allotted for each of the first two operators and 250 scooters allotted for each for the last two operators.⁸¹ Austin, Texas, has also adopted a similar approach, yet does not restrict the number of operators, only restricting the number of scooters allowed per operator (500 units per operator).⁸²

3.1.4 Absolutely Not! Temporary E-Scooter BANS

While some cities have chosen to allow these devices, others have chosen to do the exact opposite by adopting outright bans such as in Milwaukee. West Hollywood, and Beverly Hills, at least on a temporary basis.83 The one exception is that West Hollywood allows scooters

3.2 E-Scooter Safety Regulation CONSIDERATIONS: RIDING BEHAVIOUR

to the wearing of helmets.

Comparing helmet regulations in different states reveals different rules of enforcement altogether, where riders are either required or not required to wear helmets. For example, legislation in California required e-scooter riders to wear helmets, yet with the latest law, A.B. 2989, the state no longer requires adult riders (18 and over) to wear them.85 Compared to California, the State of Oregon dictates that helmets must be worn while riding scooters.⁸⁶ Additionally, in Portland, Oregon, riders can be fined for not wearing helmets.⁸⁷ Helmet laws which are meant to protect riders do have one shared quality in all cities:

to travel through their city, yet e-scooters are not permitted to be left in the area.⁸⁴ A key characteristic, regardless of the method of regulation, is that whatever measures cities have adopted, they are temporary and form part of a pilot program. The relevance of this finding points towards the flexibility of pilot programs and their ability to allow for experimentation with different methods of regulating e-scooter share.

While cities have adopted regulation to control how e-scooters operate, there are different characteristics that warrant further investigation. Part of this behaviour directly impacts safety regulation when it comes

low compliance. In Santa Monica, rider compliance with helmet laws is low. estimated to be around approximately two percent.⁸⁸ This is also the case in Portland.⁸⁹ This phenomenon is not new, and is also the case for Seattle. Washington's dockless bike program, with low compliance despite laws mandating helmet use.⁹⁰ The relevance of this finding is that whether or not the law regulates the use of helmets while riding e-scooters, compliance is low.

3.3 OPERATIONAL CHARACTERISTICS IN SIMILAR MODES OF MOBILITY AND PLANNING

Operational characteristics have been discussed in planning for other modes of micro-mobility. A 2004 study commissioned by the Federal Highway Administration (FHWA) measured the operational characteristics of several lowspeed devices with the goal of capturing data to inform the design and building of transportation facilities that meet their needs.⁹¹ Devices they studied included human-powered kick scooters, in-line skates, skateboards, human-powered and motorized mobility devices for the disabled, and Segways (often referred to in regulations as electronic personal assistive mobility devices [EPAMDs]).92 Segway devices that can perform usercontrolled stops were cited as having short stopping distances than the rest of the observed devices.⁹³ Furthermore, the

State of California permits the operation of FPAMDs on sidewalks.⁹⁴

If e-scooters have similar operational characteristics to existing-regulated devices, that could point to regulatory approaches that could be applied to scooters. Table 1 identifies speeds of other modes which may be relevant comparisons for e-scooters, as they either serve similar purposes or could potentially share space with e-scooters.

3.4 MEASURING THE **OBSERVATIONAL CHARACTERISTICS OF** MICRO-MOBILITY DEVICES

A total of twelve studies were reviewed to identify methods used for measuring the operational characteristics of different modes of micro-mobility. These include video recording equipment, direct site observation, conducting surveys and/ or questionnaires, use of on-board Global Position System (GPS) recording equipment, and use of smartphone-based applications.

3.4.1 VIDEO RECORDING

Regarding the use of video recording equipment, six of the studies focused on video recording observations. Of these five studies, both De Waard et al. (2010) and Landis et al. (2004) focused on bicycles, Miller et al. (2008) focused on Segways, Birriel et al. (2008) focused on in-line skating, and Finnis and Walton

TABLE 1. SPEED OF PEDESTRIANS. CYCLISTS. AND OTHER MICRO-MOBILITY DEVICES

	Range		
Μοσε/Device	Average Speed	(15th-85th Percentile)	Source
Pedestrians (younger, 13 to 64)	2.8 mph (4.5 kph)	-	Knoblauch, Pietrucha, and Nitzburg, 199695
Pedestrians (older, 65+)	3.4 mph (5.4 kph)	-	Knoblauch, Pietrucha, and Nitzburg, 199695
Bicycles	10.6 mph (17 kph)	7.0 – 13.7 mph (11 to 22 kph)	FHWA, 200497
In-line skates	9.86 mph (15.9 kph)	7.13 -12.59 mph (11.5 - 20.3 kph)	Birriel, Pernia, Lu, and Petritsch, 200198
Kick scooters	7.5 mph (12 kph)	5.6 – 9.3 mph (9 – 15 kph)	FHWA, 200499
Skateboards	9.7 mph (15.6 kph)	8.0 – 11.4 mph (12.9 – 18.3 kph)	Fang and Handy, 2017 ¹⁰⁰
Electric bicycles (E-Bikes)	13.6 mph (22 kph)	-	Dill and Rose, 2012 ¹⁰¹
Electric Personal Assistive Mobility Devices (e.g. Segways)	9.3 mph (15 kph)	8.7 – 10.6mph 14 – 17 kph)	FHWA, 2004 ¹⁰²

(2008) focused on walking. Of note, only Landis et al. (2004) and Lin et al. (2008) measured multiple modes of mobility. Video recording of operational characteristics of e-scooter share has been successfully used to measure comparable modes of micro-mobility that mimic e-scooter share. However, this is not to say that video recording is not without its faults. One of its limitations is the potential for those individuals being observed to alter their behaviour and act differently, as identified by both De Waard et al. (2010) and Finnis and Walton (2008).¹⁰³ Out of the six studies using video cameras, only one, Finnis and Walton (2008), prescribed a methodology for trying to reduce the incidence of altered riding behaviour: hide or obscure the camera from participants to make it less noticeable.¹⁰⁴

Regarding the more obscure methods 3.4.2 SITE OBSERVATIONS. GPS. SURVEYS. AND of recording observations, since this SMART PHONE APPLICATIONS research effort focuses on a newer On the other hand, of the studies that mode of mobility, the use of GPS units did not use video recording methods for is questionable as it would require an measuring operational characteristics, two agreement with e-scooter-share operators used volunteers to record measurements to request this information and could lead on-site: Fang & Handy (2008), and to potential lengthy conversations limiting Knoblauch, Pietrucha, and Nitzburg the time window for data collection. As (1996).¹⁰⁵ The rest of the studies used it pertains to questionnaires and surveys, more obscure methods such as Broach. this method would require volunteers. Dill and Gliebe's (2012) method of but that could also lead to bias due mounting GPS units to bicycles, Aultmanto the tendency for self-reported data Hall and La Mondia's (2005) and Weinert to be inaccurate. Moreover, there is a et al.'s (2007) use of either questionnaires potential for lower response rates due or surveys, and Ai et al.'s (2008) use to the fact that e-scooters are relatively

of a cell phone-based application.¹⁰⁶ Of these studies, the majority focused on either walking or bicycling, while Fang and Handy (2008), and Ai et al. (2018) focused on making comparisons between at least two modes of mobility.¹⁰⁷ In sum, these findings point to the potential use of video recording of multiple modes of mobility, and potentially e-scooter share. However, unlike video recording, Fang and Handy's (2008) and Knoblauch, Pietrucha and Nitzburg's (1996) method does not have to deal with the issue of altered subject behaviour due to the subjects not being aware that they are being observed. The other three studies all relied on volunteer participation prior to the start of observational measurements. thus they are still subject to participants altering their actual or reported behaviour

new, and rider ownership rates might be lower than that of bicycle owners, as noted in Weinert et al.'s (2007) study.¹⁰⁸ Additionally, Aultman-Hall and LaMondia (2005) noted that there is a potential for bias due to the over-representation of one particular group due to conducting on-site surveys and questionnaires at specific locations.¹⁰⁹ While Ai et al.'s (2008) method of using phone-based applications could overcome the problem of human error associated with selfreporting observations, it faces the same limitation regarding ownership – bicycle users can own their vehicles unlike the riders of e-scooter share.¹¹⁰

3.5 OPERATIONAL CHARACTERISTICS FOR MEASUREMENT: SPEED

Concerning speed, ten of the twelve studies reviewed looked at measuring speed.¹¹¹ Of the studies that looked at speed, only four of them looked at measurements applicable within the United States, including Fang and Handy (2008), Landis et al. (2004), Birriel et al. (2001), and Miller et al. (2008).¹¹² While each study measured speed, they utilized different methods. Where Fang and Handy (2008) used a stopwatch and a predefined distance over different transportation facilities, the other three studies used video recording equipment to later record riders traveling over a specified distance and tabulate the speed result. Unlike Fang and Handy's (2008) study, the other three studies

used obstacle courses or marked locations (visible to participants) designed ahead of time, versus the former's method of picking a particularly busy spot and recording observations. Other considerations such as Miller et al.'s (2008) study noted that the length or distance of an observation zone has an impact regarding the measurement of speed. Additionally, Miller et al. (2008) noted that their study was hindered by the Internal Review Board's speed safety regulations, effectively limiting the potential to measure the true maximum operational speed of a Segway device.¹¹³

This research effort also seeks to answer how these characteristics are relevant to the field of urban planning. Of the studies mentioned in this section, several of them justified the need to measure speed in order to improve the planning and design of transportation facilities. For example, Birriel et al. (2001) noted that these characteristics could be used to derive better design guidelines for mixed-use paths if one knows just how guickly one travels.¹¹⁴ Moreover, Landis et al. (2008) noted the need to design for "emerging" modes in order to safely accommodate all users and not just bicycles.¹¹⁵ Miller et al. (2008) also noted that characteristics such as speed can facilitate a rational discussion regarding the integration of Segway devices in terms of "regulation, planning, and designing" on mixed-use paths and streets with

other modes of mobility.¹¹⁶ Additionally, Fang and Handy (2008) made a point about the compatibility of bicycles and skateboards by comparing their speeds, and postulating that in instances in which the measurements are similar, it may not make sense to regulate these modes differently.¹¹⁷

3.6 DATA ANALYSIS REDUCTION: AGE, GENDER, AND TRAVEL BEHAVIOURS (CONFLICTS)

Another critical component of the literature revealed further criteria used to link findings from observations in order to establish trends or patterns. This method was often referred to throughout the literature as analysis reduction. When it comes to developing criteria for analysis reduction, this research effort looked at previous studies and found several common metrics such as age, gender, rider experience, and travel behaviours (such as conflicts with other users, and rider distractions). Of the twelve studies reviewed, only six of them covered the metrics mentioned above, including Miller et al. (2008), Fang and Handy (2017), De Waard et al. (2010), Birriel et al. (2001), Lin et al. (2008). Knoblauch. Pietrucha. and Nitzburg (1996), and Finnis and Walton (2008). Lin et al. (2008) noted that age is important to measure because it impacts speed based on a rider's physical attributes such as strength, and the rider's mental state such as the ability to

make risky decisions.¹¹⁸ Most importantly, the authors of these papers used analysis reduction metrics to determine any correlations among operational characteristics such as speed.

Miller et al. (2008) chose to use Segway riders' level of experience to categorize their findings for speed measurement.¹¹⁹ Miller et al. (2008) found experienced riders traveled at an average speed that was faster than non-experienced riders.¹²⁰ Note that this study was able to segregate riders by experience level through a prescreening questionnaire.¹²¹ While Birriel et al. (2001) also used rider experience (related to in-line skaters), they used site observations and gender to gauge this metric.¹²² Birriel et al. (2001) found women traveled at lower speeds than men.¹²³

Unlike the previous two studies, in addition to recording gender, Fang and Handy (2017) used distraction conflicts to make comparisons among skateboarders and bicycle riders.¹²⁴ Fang and Handy (2017) focused on observing skateboarder "traffic conflicts" with other users by recording whether they rode in a straight line or evaded the conflict by swerving out of the way.¹²⁵ Additionally, Fang and Handy (2017) recorded "distraction conflicts" about multi-tasking such as eating and cell phone use.¹²⁶ While Fang and Handy (2017) found that men traveled faster than women on bicycles when compared to skateboards, no findings among gender were made for skateboarders due to the over-representation of males (97 out of 100) among observed skateboarders.¹²⁷ Concerning distraction conflicts, few were found among skateboarders.¹²⁸

De Waard et al. (2010) also looked at distraction conflicts but focused on bicycle riders' age, gender, and the number of hands on bicycle handlebars.¹²⁹ In this study, the authors cited secondary tasks as distraction conflicts such as cell phone use, listening to music, and chatting with another bike rider.¹³⁰ Concerning age categories, the authors used four classifications: less than 15, 15 to 30, 30 to 50, and 50 plus.¹³¹ De Waard et al. (2010) found a tiny percentage of riders were distracted while riding, similar to Fang and Handy's (2017) study.¹³² However, De Waard et al. (2010) did establish a link between gender and handlebar use, where more women than men were founding riding with one hand on their bicvcles.¹³³

Knoblauch, Pietrucha and Nitzburg (1996), Finnis and Walton (2008), and Lin et al. (2008) also measured apparent gender and age as did De Waard et al. (2008), but not all authors used the same levels of age classifications. Knoblauch, Pietrucha and Nitzburg (1996) focused strictly on pedestrian walking speeds for people 64 years and younger or 65 and

older. Their study showed adolescent males walked faster than older females. and people who traveled in groups walked slower than those walking by themselves.¹³⁴ Finnis and Walton (2008) also measured pedestrian walking speeds but followed the exact age classifications as De Waard et al. (2010).¹³⁵ Finnis and Walton shared similar results to Knoblauch, Pietrucha, and Nitzburg's (1996) study where men walked faster than women while older individuals walked slower than younger individuals.¹³⁶ Note, Finnis and Walton (2008) did not measure group travel among pedestrians.¹³⁷ Lin et al. (2008) used three age categories for observations. less than 25, 25 to 50, and 50 plus.¹³⁸ Lin et al. also found similar results to Finnis and Walton's (2008), and Knoblauch, Pietruch and Nitzburg's (1996) studies, that bicycles riders also exhibit similar traits: males travel faster than females, and vouths travel faster than older adults.¹³⁹

3.7 Why the Scoot Should Cities Care

In summary, this literature review answered several questions of the e-scooter share debate concerning regulation, the role of operational characteristics in planning in similar modes of micro-mobility, measurement methodologies for operational characteristics and riding behaviours, and analysis reduction metrics. These findings would serve as a basis for developing a testing methodology as discussed in Chapter 4.

There is an apparent disagreement among local governments over where these devices should operate, pointing to compatibility issues between e-scooter share riders, pedestrians, and other similar modes of mobility sharing the same space. Different cities across the United States have regulated e-scooter share to segregate e-scooter riders to either streets or a mix of both streets and mixed-use paths. Sidewalk operation appears to be the most contentious issue as seen in Santa Monica, Los Angeles, and San Francisco. CA: however, there was an exception: Denver, CO, In Denver, e-scooters can only operate on sidewalks. On the other hand, governments have also tried to regulate these devices by using speed, limiting the number of operators and units, or by outright banning e-scooters, at least on a temporary basis.¹⁴⁰ While governments disagree on where and how to regulate, they also disagree on safety regulation of e-scooter share as it pertains to the wearing of helmets. State governments in Washington and Oregon require scooter riders to use helmets.¹⁴¹ California backpedaled and removed this requirement for adult riders.¹⁴²

With so much disagreement, what alternatives exist to guide the development of the design and regulation of transportation facilities accommodating

e-scooter share? The literature review revealed that the City of Los Angeles uses speed, an operational characteristic, to regulate e-scooter share. Furthermore, the most common methods to measure operational characteristics are using a camera to film riders or by manually observing and recording riders. While both methods have their drawbacks, this research effort followed Fang and Handy's (2008) and Knoblauch, Petri and Nietzburg's (1996) method of using on-site recording to measure operational characteristics rather than using a camera. Using site observations instead of video recording methods reduces the possibility of individuals altering their riding riding behaviour characteristics which will behaviour, helps avoid potentially lengthy discussion regarding privacy, and dealing with low response rates or inaccuracies associated with self-recording.¹⁴³

The literature also revealed the relevance of measuring operational characteristics. In California, the fear over scooters traveling at 25 mph led the government to remove sidewalk operation from the bill.¹⁴⁴ This perception highlights an issue that is central to this project: how can one know whether this fear is founded in truth in the absence of data collection? Additionally, operational characteristics such as speed could be used to derive better design guidelines for different transportation facilities if one knows how quickly the average user travels.¹⁴⁵

While measuring the operational characteristics of different modes of mobility is important, how is it possible to make it relatable to people? The studies summarized in this chapter revealed that analysis reduction metrics such as age, gender, and travel behaviours (e.g. concurrent cell phone use, travel conflicts) can be used to make associations with operational characteristics. Linked together, both operational characteristics and analysis reduction metrics can aid in establishing trends or patterns that help to develop an understanding for these relationships. Additionally, these metrics help fill a void in the literature pertaining to facilitate a discussion around the rational planning and regulation of transportation facilities, including e-scooter share.

Chapter 4 will build upon the findings from the literature in order to establish a testing methodology for measuring the operational characteristics and rider behaviour of e-scooter share.







E-scooter riders were observed riders. Observations were conducted from in downtown San Jose, California. October 2018 to February 2019. The methods used for measuring operational Observations were based on speed and other safety-related rider behaviours. characteristics are derived from previous Three types of facilities were analyzed: studies conducted on similar modes of streets, sidewalks, and mixed-use paths mobility referenced in Chapter 3. closed to vehicular traffic. E-scooters were observed on streets and sidewalks on 4.1 SPEED Santa Clara Street from Almaden Avenue Regarding speed, observations focused on measuring the "space-mean speed" were carried out on a pedestrian and defined as "the average speed of bicyclist mixed-use paths section of vehicles traveling on a given segment of Seventh Street through the San Jose State roadway during a specified period and is University campus, adjacent to the Student calculated using the average travel time and length for the roadway segment" by the Federal Highway Administration's Locations were chosen based on previous (FHWA) Travel Time Data Collection pilot observations conducted over the Handbook.¹⁴⁶ To calculate speed, the summer of 2018 to determine locations elapsed time it took for an e-scooter

that exhibit a high number of e-scooter

MERIOOS

CHAPTER 4

rider to travel between two checkpoints over a defined distance was measured. Similar methods were used by FHWA (2004), Birriel, Pernia, Lu, and Petritsch (2001), and Fang and Handy (2017), to measure speed of multiple mixed-use trail users, in-line skaters, and skateboarders, respectively.¹⁴⁷

Collecting speed observational data helps answer just how quickly e-scooter streets, and mixed-use paths. In order to establish a comparison against similar modes of micro-mobility, bicycle riders were observed on the street facility only. Speed results were used to identify any trends observed on different facilities and form the basis for a comparison against similar modes of micro-mobility.



Time was also measured to determine how long it took a rider to cross the observation zone in real time using a smartphone (iPhone X) stopwatch application. Checkpoints in the various observation locations included existing. easily identifiable objects present in the streetscape including light poles, sign poles, trees, other street furniture, and pavement markings. Since pre-existing street furniture and markings were used in the various speed observation zones, the length of the speed observation zones varied between approximately 121 and 125 feet (36.9 - 38.1 m) in length (see Figures 13 and 14). In addition to the travel time, the date, facility type. temperature, recorded time, e-scooter operator (Lime or Bird), rider's apparent gender, apparent age, style of travel, helmet use, number of riders, conflicts, and distractions were also recorded. All observations were recorded on paper and later transferred to Excel.

Concerning the analysis of speed results, several inferential statistical tests were used to determine the significance of the speed results. An Excel two-sample t-test assuming equal variances was used

In total, 330 e-scooter riders were observed for speed, including 110 on the street segment, 110 on the sidewalk segment, and 110 on the mixed-use path segment. Moreover, 110 bicycle riders were observed on the street facility only.

to determine the statistical significance when comparing street vs. sidewalk speeds, street vs. mixed-use path speeds, sidewalk vs. mixed-use path speeds, apparent gender speeds, operator speeds, and mode speeds (e-scooter versus vs. bicycle). An Excel single factor ANOVA test was used to compare e-scooter speeds among all three transportation facilities (streets, sidewalks, and mixeduse paths), apparent gender speeds (both male and female) across all three transportation facilities, and age categories and speed. Simple linear regression analysis in SPSS was used to examine the relationship between temperatures and speed.

4.2 RIDING BEHAVIOURS

In addition to the speed, rider behaviours were recorded that may potentially influence safety. Behavioural observations took place in the same locations as speed, including street segments, sidewalk segments, and mixed-use path segments

4.2.1 HELMET USE

Helmet use is a key safety indicator of rider behaviour that this report examined among e-scooter riders, expressed as a percentage, with the number of helmet users observed divided by the total number of observations.

This is particularly interesting in the study in San Jose, as riders not wearing helmets under the age of 18 are in

violation of the current California Vehicle Code.¹⁴⁸ Removing that requirement for bicycle helmets for adults is under consideration in the latest draft of the state A.B. 2989 legislation. Helmet use findings may have implications that will play a role in the determination of future e-scooter regulation. Note that while this helmet requirement was removed with the passing of A.B. 2989, it is still illegal for those under the age of 18 to travel without a helmet. Additionally, there are growing concerns over e-scooter injuries as a result of riders not wearing helmets.149

4.2.2 TRAVEL STYLE: STRAIGHT FORWARD OR SIDE TO SIDE

Additionally, travel style was observed, specifically whether e-scooter riders travel in a straight line or if they traveled in an irregular, less predictable "carving" motion, traveling in a S-shaped style motion from side to side.¹⁵⁰ Travel style findings may have implications that will play a role in the determination of future e-scooter regulation as carving proves to be a contentious issue over the regulation of similar modes of micro-mobility.¹⁵¹ Each riding style was expressed as straight forward and side to side, and the results were calculated as a percentage, with the number of observed riders divided by the total number of observations.

4.2.3 GROUP RIDING/MULTIPLE RIDERS

E-scooter riders that traveled solo or in



groups were also observed. The groups by either party having to slow down or were categorized as riders traveling solo swerve out of the way.¹⁵³ Additionally, where was is evident that there were the type of conflict was also recorded, no other e-scooter riders around them. such as a pedestrian, another e-scooter, Group riding was classified as those who a cyclist, a parked car, or moving travel in groups of two or more individuals vehicular traffic. Traffic conflict findings traveling in apparent "packs" together. To may have implications that will play a role differentiate a pack from two independent in the determination of future e-scooter riders who happen to be riding alongside regulation as it pertains to the interaction each other, evidence of users interacting between e-scooters and other users on with each other during travel was used to streets, sidewalks, and mixed-use paths. make that distinction.

Additionally, instances of different types of group riding were recorded; namely, two individuals sharing the same e-scooter. Pilot observations carried out in summer of 2018 revealed that riders traveled in groups or with multiple riders on one e-scooter. The potential implication of these observations may play a role in the development of e-scooter regulation operation, infrastructure design, and e-scooter design itself.

Group and multiple riders' results were expressed as a percentage, with the number of observed riders divided by the total number of observations.

4.2.4 TRAFFIC CONFLICTS

Similar to Fang and Handy's (2017) study of skateboarders, traffic conflicts were also observed.¹⁵² Traffic conflicts are defined as situations where riders encounter obstacles such as pedestrians or other vehicles and cause them to react

Marmons

Traffic conflicts were expressed as a percentage, with the number of observed riders divided by the total number of observations.

4.2.5 RIDER DISTRACTIONS

There are several activities other than operating a scooter that a rider could engage in that could potentially distract a rider, such as the use of headphones or smartphones.

E-scooter riders were observed for several types of multi-tasking including the use of smartphones and the wearing of headphones. Distractions warrant further investigation as the wearing of headphones is not prohibited for e-scooter riders, but bicyclists can only have a headphone in one ear.¹⁵⁴ Additionally, collecting this information allows for comparisons against other modes of micro-mobility. De Waard, et al. (2010) found approximately 2.2 percent of bicyclists talking on their cell phones.¹⁵⁵ Fang and Handy (2017) found that 3 percent of skateboarders were making phone calls while riding.¹⁵⁶ Potential distraction findings may have implications that will play a role in the determination of future e-scooter regulation.

Each rider distraction was expressed as a percentage, with the number of observed riders divided by the total number of observations.

4.3 ENVIRONMENTAL CONSIDERATIONS: WET. DRY WEATHER. AND TEMPERATURE

As it pertains to weather considerations, riders were observed in a mix of both wet and dry weather conditions. Moreover, temperature was also recorded, which ranged between 39- and 78-degrees Fahrenheit (4 to 26 degrees Celsius).

4.4 ANALYSIS REDUCTION METRICS: AGE. APPARENT GENDER

Riders of different ages were observed and categorized into three groups: those younger than 25, those 25 to 50, and those over 50 years of age. Concerning apparent gender, riders were classified as either male or female. The identification of age groups and apparent gender was determined through observations by the observer.

4.5 METHODOLOGY SUMMARY

In summary, this chapter established a testing methodology in order to measure the operational characteristics and riding behaviour of e-scooter share riders in San Jose on three types of facilities: sidewalks, mixed-use paths, and streets. The chosen method involved using onsite observations inspired by the literature review in Chapter 3.

Speed or the "space mean speed" was calculated by measuring the time it took to cross one of these facilities over a predefined length in feet.¹⁵⁷ At the same time, riding behaviour along with traffic conflict observations were observed in the same locations as speed measurements. Additional measurements such as apparent age, gender, and temperature were included in order to better understand any trends or patterns.

Chapter 5 will review the findings of these observations along with any potential trends or patterns linked to reduction metrics or environmental considerations.







This chapter explores the results from obtransportation facilities (streets, sidewalks, and mixed-use paths). Speed results cover several topics, including transportation facilities, ranges, gender, operator, age, scooters versus bicyclists, and temperature. Rider behaviour results cover riders, distractions (cell phones and headphones), traffic conflicts, and riding styles (straight forward or side to side), and rider collisions. An additional 110 bicycle riders to compare speed, gender, and helmet

CHAPTER 5 NOT SO FAST NOR DISTRICTED

5.1 RIDERS TRAVEL FASTER ON STREETS. SLOWER ON SIDEWALKS. BETWEEN 9.0-11.1 MPH

Rider speed varied on the three types o transportation facilities studied, with many of the differences resulting in statistically significant results. Riders traveled on aver age 11.1 mph (17.9 kph) on streets, 8.9 mph (14.5 kph) on sidewalks, and 9.6 mph (15.5 kph) on mixed-use paths (see Figure 15). In other words, riders traveled faster on streets and slower on sidewalks and mixed-use paths. Importantly, riders slowed down when traveling on facilities mixed with pedestrian traffic. Furthermore comparing the two pedestrian facilities, riders traveled faster on the wider mixeduse path versus the narrower sidewalks.

5.2 WIDE RANGE OF RIDER SPEED BETWEEN 4 AND 16 MPH

As it pertains to the distribution of speeds among all 330 riders observed, there was wide variation (see Figure 16). The lowest measured travel speed was 4.0 mph (6.4 kph), similar to that of a pedestrian walk-16.1 mph (25.9 kph).

This high speed is somewhat perplexing, as this is above the speed by which e-scooter share operators claim to limit their e-scooters. Overall, seven scooters (2 percent) exceeded the 15-mph speed limitation, while 50 scooters (15.2 percent) exceeded San Jose's current ordinanceimposed limit of 12 mph.¹⁵⁸ E-Scooters exceeding 12-15 mph could be due to



a number of reasons. There could be an error in measurement technique, as it pertains to speed, due to either time calculations or measurement of distance between observation zone checkpoints. However, measurements would have had to be off by several seconds or tens of feet to make a difference, which is unlikely given the distances and lengths of time of individual observations. Alternatively, there could be a margin of error in how the devices limit speed. Also, it could be that devices limit speed by restricting motor power to a certain level that theoretically produces a certain speed rather than capping the top maximum speed. Additionally, it is a possibility that the target maximum speed could

be exceeded in certain environmental conditions, such as a downslope in the direction of travel or a strong tailwind pushing the rider. Additionally, concerning San Jose's e-scooter speed ordinance, while approved by council in December of 2018, it was not enacted until after February of 2019, after site observations were completed.¹⁵⁹

In general, most riders are traveling well below the maximum mechanical speeds of e-scooters. Among all 330 riders observed, regardless of location, the average speed was 9.9 mph (16.0 kph). Sixty-seven percent of riders were below 11.0 mph (17.7 kph).

5.3 MALES RIDE FASTER. VARY Speed Less by Facility

As it pertains to apparent gender, there was a significant difference in terms of the mean speed between males and females (p < 0.01). Female riders traveled at a lower average speed than males, 9.3 mph (15.0 kph) for females, and 10.1 mph (16.3 kph) for males (see Table 2). When comparing the speeds of apparent genders to similar modes of micromobility, similar results were observed among in-line skaters and bicyclists. Birriel et. al (2001), found that male inline skaters were faster than females. while Fang and Handy (2017) found that male cyclists rode faster than females.¹⁶⁰



FIGURE 16. DISTRIBUTION OF E-SCOOTER RIDER SPEEDS ON DIFFERENT TYPES OF TRANSPORTATION FACILITIES.





Furthermore, this study found variances when comparing gender and type of transportation facility (see Table 2). For both genders, riders were faster on streets than on mixed-use paths and sidewalks. However, the variance of the speed difference by facility was more pronounced among female riders. When compared to males, it appears that female riders slowed down more in the presence of pedestrian traffic. Additionally, female riders tended to travel faster on streets than male riders. Interestingly, female riders were 29 percent slower when traveling from streets to sidewalks, while male riders were just 23 percent slower (see Table 2). Ultimately, this finding is interesting because it appears that female riders tend be much more courteous

riders than male riders (by slowing down more than men) when riding on facilities where pedestrians are present. Of note, San Jose's male versus female population is evenly split with 50 percent of the population identifying as male, and 49.7 percent of the population identifying as female.¹⁶¹

5.4 No Difference by E-Scooter Share Operator

No observed differences in speed were observed between Lime and Bird e-scooters. No e-scooters from recent entrants Skip or Wind were seen during the observation period. Riders that used Bird traveled at similar average speeds to those that used Lime e-scooters, 10.0 mph (17.9 kph) versus 9.9 mph (17.2 kph) respectively, but the slight difference was not significant (p=0.68). Interestingly, when looking at the speed distribution between Bird and Lime, visually there is less variance among Bird riders such that a higher proportion of observations are concentrated between 7 to 13 mph (see Figure 17). Lime riders show a greater variance in terms of speed amongst the entire distribution of measured observations. Of note, Lime scooters feature electronic speedometers, while Bird scooters do not.

Furthermore, Bird and Lime use different scooter models. Bird uses Xiaomi–Mi e-scooters which weigh approximately 26.9 (12.2 kg) pounds and feature

TABLE 2. DIFFERENCES IN SPEED BY FACILITY AND GENDER

Speeds By

Sidewalk

Mixed-use par

Street

Overall

PERCENT II FROM...

Sidewalk to m

Mixed-use pa

Sidewalk to str

PERCENT D

Street to mixed

Street to Sidew

Comparison: t-test (female

LOCATION (MPH)	Female (n=79)	MALE (N=251)
	8.0	9.3
th	9.1	9.9
	11.3	11.1
	9.3	10.1
ICREASE IN SPEED	Female	MALE
ixed-use path	14%	6%
th to street	24%	12%
reet	41%	20%
ECREASE IN SPEED	Female	MALE
d-use path	17%	11%
walk	29%	23%

t-test (female vs. male, overall): p =<0.01.

a motor power of 250 watts.¹⁶² Lime currently uses the latest version of scooter from Segway–Ninebot weighing approximately 28 pounds (12.5 kg) with a motor rated power of 300 watts.¹⁶³

5.5 Older Riders Travel Faster Than Their Younger Counterparts

As noted in Chapter 3, age is an important characteristic to consider as it can impact a person's travel speed based on their physical ability.¹⁶⁴ Observed e-scooter riders were categorized into three age groups: adolescent riders younger than 25 (n=106), adults between the ages 25 to 50 (n=192), and older adults over 50 years of age (n=32). The results (see Figure 18) show a statistically

significant finding (p<0.01) in terms of speed and age, where each group traveled at a different average speed. Younger (adolescent) riders traveled at an average speed of 9.4 mph (10.1 kph), those adults between the ages of 25 to 50 traveled at 10.6 mph (17.1 kph), and older adults over the age of 50 traveled at 10.4 mph (16.7 kph). Based on these observations, the results point to an interesting finding: that adults and older riders traveled slightly faster than adolescents. Of note, San Jose's population skews towards the second age bracket of adults with a median age of 35.4 and with the majority of the population falling between the ages of 18 to 64 (65 percent).¹⁶⁵

Comparing e-scooter speeds based on age to that of similar modes of mobility, studies revealed an opposite trend. For example, compared to bicyclists, Lin et al. (2008) found that adolescents rode faster than adult bicyclists.¹⁶⁶

5.6 E-Scooter Riders Travel Slower than Cyclists

Previous studies mentioned in Chapter 2 that have measured speed in similar modes of micro-mobility have also measured other modes in order to establish comparisons between the two such as Fang and Handy (2017), and Lin et al. (2008).¹⁶⁷ In this study, when comparing e-scooter speeds to bicycle speeds, the results show a



FIGURE 18. DISTRIBUTION OF E-SCOOTER RIDER SPEEDS BY AGE IN YEARS (LESS THAN 25, 25 TO 50, AND 50 AND OVER).



statistically significant finding (p<0.01), where e-scooter riders traveled on average about 11.1 mph (17.9 kph), versus cyclists who traveled at 12.2 mph (19.6 kph). In other words, e-scooter riders traveled slower than cyclists. Approximately half of all observed cyclists traveled at an average speed ranging between 9 to 13 mph (14.4 to 20.9 kph). Of note, cyclists (n=110) were only observed riding on streets for this comparison. Similarly, Fang and Handy (2017) found that skateboarders just like e-scooters traveled at slower speeds when compared to cyclists (on average 9.7 mph compared to 11.6 mph).¹⁶⁸

5.7 TEMPERATURE VERSUS SPEED

As it pertained to e-scooter riders and the ambient temperature, some variances were observed in terms of speed due to changes in temperature. After running a simple linear regression in SPSS to determine the impact that temperature had on speed, the results showed a statistically significant finding (see Figure 20). The results indicated that as temperature increased, e-scooter speeds decreased (p=0.04). While this result is statistically significant, the R² value was small, indicating there were other factors that could have contributed to the variance in speed that were not measured as part of this study.

5.8 MORE CYCLISTS WEAR HELMETS THAN E-SCOOTER RIDERS

Sidestepping the issue of whether wearing a bicycle helmet should be regulated, it is no longer California State law that e-scooter riders are required to wear them, except for those under the age of 18.¹⁶⁹ E-scooter share mobile applications inform users of this fact during the signup process. Acknowledging this is a requirement based on age, very few incidents were observed. Only seven incidents, or two percent, out of a total 330 observations were observed where scooter riders wore helmets (see Figure 21). Notably, six of these observations



FIGURE 20. DISTRIBUTION OF E-SCOOTER RIDER SPEEDS ORDERED BY TEMPERATURE RANGES.



FIGURE 21. PERCENTAGE OF E-SCOOTER RIDERS AND CYCLISTS THAT WEAR HELMETS. PHOTOGRAPHY BY THE AUTHOR.

occurred on the street, while only one occurred on the sidewalk. This low compliance value is similar to that of Trivedi et al.'s (2019) study on e-scooter safety in Los Angeles where only 5.6 percent of observed riders observed wore helmets.¹⁷⁰

Additionally, none of the helmet wearing scooter riders were part of the younger population (less than 25 years of age), indicating the possibility that some riders may be in contravention of California State law. Comparatively, observing cyclists (n=110), more than half of all observed cyclists wore helmets, approximately 56 percent (see Figure 21). Moreover, out of the 60 observations where cyclists were found to be wearing helmets, 72 percent (n=43) fell into the younger adult population (25 years to 50), while 28 percent (n=17) were older adults over

the age of 50. Of note, cyclists that wore helmets were observed using their own personal bicycle instead of a bike sharing services such as San Jose's Ford Go Bike.¹⁷¹

5.9 A NOTICEABLE AMOUNT OF GROUP RIDING

E-scooters are sometimes described as a type of "personal transportation device", providing mobility for individual travelers. Thus, it is not surprising that a majority of riders observed were traveling solo, approximately 83 percent (n=273). However, there was a noticeable number of riders traveling in packs. Out of the 330 observations made, approximately 16 percent of riders were found traveling in groups, with 11.5 percent in groups of two, 4.5 percent traveling in groups of three, and one apparent pack of five

riders (see Figure 22).

Eleven incidents (3 percent) in which two riders would share one e-scooter were observed. The operator of the e-scooter would stand in the front grabbing onto the handle bars while the second individual would stand behind the operator either holding onto their shoulders or waist (see Figure 23). Comparing San Jose to Los Angeles, 7.3 percent of e-scooter riders in Los Angeles were found to be sharing an e-scooter.172

5.10 MINIMAL RIDER DISTRACTION: Some Wear Headphones, Very Few USE CELL PHONES

Travelers being distracted, particularly by cell phones, is often cited as a problem for many modes of transportation.



FIGURE 23. PERCENTAGE OF MULTIPLE RIDERS SHARING ONE E-SCOOTER. PHOTOGRAPHY BY THE AUTHOR.

Interestingly, only one rider was observed holding a cell phone while riding an e-scooter (see Figure 24). Even then, it was observed that the rider still slowed down significantly in order to adjust their balance and place their body forward on the handle bars. This finding is similar to that of bicyclists observed using cell riders were also showing significant

Additionally, 53 incidents (16 percent) phones by De Waard et al. (2010) where were observed where riders were wearing headphones (see Figure 25). decreases in speed.¹⁷³ Depending on what those headphones are connected to, they could be listening For comparison, De Waard, et al. (2010) to music off a smartphone or making a found approximately 2.2 percent of phone call. In the study area, it would be bicyclists talking on their cell phones.¹⁷⁴ illegal for a bicyclist to wear headphones Fang and Handy (2017) found that 12 in both ears according to state law.¹⁷⁶ percent of skateboarders were holding Comparing e-scooters riders to bicyclists. cell phones riding.¹⁷⁵ Essentially, e-scooter Wolfe et al.'s study found approximately 17.7 percent of bicyclists were using riders are less distracted by cell phones headphones while riding, higher than the when compared to these other modes. 7.7 percent of bicyclists observed by De This could be attributed to the fact that an e-scooter requires a user to balance the Waard et al. (2010).¹⁷⁷



Nor So Fast Nos Distance



FIGURE 22. PERCENTAGE OF E-SCOOTER RIDERS TRAVELING SOLO OR IN GROUPS.

scooter while having both hands on the handlebars. Additionally, this observation does not take into consideration whether a rider may have potentially used a cell phone at an intersection while at a crosswalk or at a red light.

5.11 TRAFFIC CONFLICTS: SLOWING Down vs. Swerving Out of the WAY

Concerning traffic conflicts, e-scooter riders were observed concerning their interactions with other users in the public right of way. These interactions included evasive actions that e-scooter riders took to avoid conflicts with pedestrians, cyclists, other e-scooter riders, and motor vehicles by either slowing down or swerving out of the way. Out of 330 observations, there were 134 traffic conflicts; 71 percent of all observed (n=95) slowed down for conflicts, while 29 percent of all riders (n=39) swerved out of the way. Of note, of the 95 incidents where e-scooter riders slowed down when encountering a conflict, 70 percent (n=66) were incidents that occurred on facilities with pedestrian mixed traffic such as the sidewalk or

PHONES. PHOTOGRAPHY BY THE AUTHOR

mixed-use path. Note, this is a positive result. E-scooter riders are yielding to pedestrians, and pedestrians are not necessarily jumping out of the way to avoid a potential collision. Moreover, of the 95 incidents where e-scooter riders slowed down for conflicts on streets (n=29), one quarter of the slowdowns were attributed to a mix of either parked cars or vehicular traffic that restricted the space which an e-scooter could travel through.

Of the 39 incidents where riders swerved out of the way to avoid a conflict, 31 of them (80 percent) occurred either on the sidewalk or on the mixed-use path, with the rest occurring on the street. Where riders encountered conflicts on the street, the majority of the incidents involved motor vehicles changing lanes or parking.

Interestingly, very few conflicts were observed with cyclists or other e-scooter riders. In total, only one incident for each of these cases was observed. In the case of the e-scooter rider conflict, on the sidewalk, both riders were observed slowing down to negotiate their path of travel. In the case of the bicycle conflict, on the street, the e-scooter rider slowed down to yield to the cyclist.

5.12 MOST E-SCOOTER RIDERS TRAVEL IN A STRAIGHT LINE

As noted by Fang and Handy's (2017) study on skateboarders, sometimes riders

were observed engaging in what could be construed as dangerous riding behaviour (performing tricks) by moving in a less predictable side to side "S-shaped" motion.¹⁷⁸ This type of riding behaviour is sometimes used as an argument for banning skateboards from a particularly facility as it might pose a problem to those around them due erratic changes in travel direction.¹⁷⁹ Similarly, this study looked at whether e-scooter riders engaged in similar riding behaviour that could also be interpreted as a potential dangerous riding activity. Approximately 97 percent of riders (n= 318) predictably travel forward in a straight line, while 3 percent of riders (n=12) rode in a less predictable pattern from side to side. Interestingly, while very few riders were observed engaging in this more casual side to side riding, incidents occurred on the mixed-use path (n=9). and most riders were under 25 years of age (n=8). Ultimately, while a few riders were observed engaging in what could be considered dangerous riding behaviour, the majority did not. This result is comparable to that of skateboarders. where the majority rode in a straight line instead of riding in a side to side motion.¹⁸⁰

5.13 Some RIDERS FALL WHILE OTHERS COLLIDE

According to Trivedi et al.'s study on e-scooter injuries, the most common types include falls and collisions with moving objects or vehicles.¹⁸¹ In this study, only one incident was observed where an e-scooter rider collided with the sidewalk curb and fell onto the ground while crossing an intersection; however, they guickly recovered and kept on riding. The other two incidents were observed on the street where riders were trying to negotiate the narrow travel path segment created by parked cars and moving vehicular traffic. In both cases, the riders collided with parked vehicles' side view mirrors sticking out in the e-scooters travel path. In both cases, the riders clipped their shoulder area. but proceeded to ride. Importantly, while this result is not a statistically significant finding, it does indicate a similar finding to that of Fang and Handy's (2017) analysis on skateboarder injuries, that vehicles pose a safety hazard for similar modes of micro-mobility on streets.¹⁸² Note that these events were not included as part of the speed tabulations and were tracked and recorded separately from other riding behaviours. In all cases, the riders were male and did not wear helmets.

No collisions with pedestrians were observed. This is an important finding as 4.4 percent e-scooter collisions in Los Angeles were reported between pedestrians and e-scooters.¹⁸³

5.14 The Results: E-Scooter Riders in San Jose

In total, 330 e-scooter riders were

SUIDAT DAIXERS, OR TOO FAST AND TOO FURIOUS?

observed in downtown San Jose during a mix of both dry and wet weather conditions across streets (n=110), sidewalks (n=110), and mixed-use paths (n=110) between the months of October 2018, and February 2019. Additionally, 110 cyclists were observed on the street facility only. Concerning average speed, riders traveled between 9 to 11 mph with a different speed per facility (p < 0.01), male riders traveled faster than females and varied less by facility (p<0.01), operator speeds were similar at 10 mph. but not statistically significant (p=0.69), older riders (10.5 mph) traveled faster than younger riders (9.4 mph) (p<0.01), e-scooter riders traveled slower on streets (11.1 mph) than bicyclists (12.2 mph) (p<0.01), and riders traveled faster in colder temperatures than warmer temperatures (p=0.04).

Furthermore, regarding riding behaviours, 2 percent of e-scooter riders wore helmets compared to 56 percent of cyclists, and the majority of e-scooter riders (96 percent) traveled in a straight line as opposed to traveling in a less predictable side to side motion. Concerning group riding and multiple riders, 17 percent of riders were observed traveling in groups while 3 percent of riders traveled with two riders on onescooter. Regarding rider distractions, only one person was seen using a cell phone. Moreover, 16 percent of riders were observed using headphones, which was

64

slightly similar to cyclists.

In terms of travel conflicts, 71 percent of riders slowed down while 29 percent swerved out of the way. Furthermore, 70 percent of riders who encountered conflicts on facilities with mixed pedestrian traffic slowed down for pedestrians (a positive finding). Of note, the few collisions observed (n=3), were as a result of a rider colliding with a curb or due to a rider colliding with a parked car's side view mirror. This finding was similar to that of skateboarder collisions.

While it is apparent that speed is dependent on many factors and riders exhibit different riding behaviours, what is not clear is the implications of these findings. Chapter 6 will conclude with a look at the potential implications.





CONCUSIONS AND IMPLICATIONS

CHAPTER 6 CONCLUSIONS AND MARZUGATIONS

While cities are instituting temporary pilot programs and regulations to deal with the influx of e-scooters, they are still in the early process of debating the compatibility following question: of these devices with other road users. This debate over whether individuals view electric scooters as compatible or incompatible with other road users and whether they support or do not support specific policies, is likely formed by how they see the attributes of e-scooters versus that of similar modes of micromobility. However, there is a lack of data on how e-scooters riders actually behave and how that compares to other road users. As a result, this research effort sought to observe e-scooters on sidewalks, streets, and mixed-use paths in San Jose with a goal of revealing how operational characteristics, riding

behaviours, and traffic conflicts could inform those debates. Specifically, this research effort sought to answer the

What are the "operational characteristics' exhibited by e-scooter share users using streets, sidewalks and mixed-use paths in downtown San Jose, California and what are the potential implications of the findings on the future development of e-scooter regulation and micro-mobility (bicycle) infrastructure design?

This study opened with a discussion about the U.S. history of micro-mobility from early bike-share systems to the explosion of e-scooter share. A literature review was then conducted to understand the regulatory debates concerning

e-scooters. Literature was also reviewed to understand methodologies behind measuring the operational characteristics. riding behaviours, and traffic conflicts in similar modes of micro-mobility. Findings were summarized concerning the importance of using operational characteristics, riding behaviours, and traffic conflicts from an urban planning perspective.

Also using a review of the literature, a 330 e-scooter riders in downtown San Jose across streets, sidewalks, and mixed-use paths. Speed, riding behaviours, and traffic conflicts were then observed and reported in Chapter 5. In this chapter, implications were derived from the findings as it pertains to potential e-scooter regulation and the design of urban infrastructure.

6.1 Not so Fast and Furious. Speeds are Similar to Other MODES

Riders are not traveling at 30mph down sidewalks. let alone on streets. On average, riders traveled faster on streets and slower on sidewalks. In particular, riders on average rode just below 8.9 mph on sidewalks. This is much less than the figure of 25 mph originally proposed as part of legalization A.B. 2989 that spawned critics of the legislation. Furthermore, e-scooter riders traveled significantly slower in the presence of pedestrians. The average e-scooter speeds on sidewalks were slightly slower when compared to what previous studies have found for in-line skaters and skateboarders. Moreover, e-scooter sidewalk speeds were similar to that of EPAMDs (e.g. Segways). All three of these other micro-mobility devices are allowed on sidewalks in California pointing to a consistency issue with the existing regulations for different modes. Overall, it appears that riders are more Sunday drivers as opposed to being too fast and too furious.

6.2 CREATING NEW CONFLICTS: SLOW SCOOTERS AND FAST CARS

Currently, the City of San Jose regulates an e-scooter vehicles' speed to a

maximum of 12 mph.¹⁸⁴ Additionally, the City also mandates that e-scooter riders must ride on the street.¹⁸⁵ However, the current speed limit for motor vehicles on Santa Clara St. between the Autumn and Market St. stretch that encapsulates one observation zone is 30 mph.¹⁸⁶ In a mixed traffic scenario such as Santa Clara St. without the presence of any micromobility infrastructure, and where you have motor vehicles that can travel or exceed the current posted speed limit, yet allow e-scooters to operate on the same facility well below the speed limit, does this not create a conflict between the two vehicles? In Santa Monica. e-scooters are limited to 20 mph. vet are expected to travel with the flow of traffic.¹⁸⁷ How do cities expect e-scooter riders to safely negotiate a lane change without the ability to accelerate to match current traffic speed conditions? Instead, cities regulate the operation of motor vehicles through speed limits on roads. It is not uncommon to regulate mixed-use paths or trails through the use of posted speed limits such as the City of Seattle implementing a 15 mph speed limit for all trail users.¹⁸⁸ In California, class I and class II e-bikes, capable of traveling up to 20 mph, are permitted on multi-use paths and trails as per California Code 212230.189 Moreover. Oakland also permits e-scooters to use the Lake Merritt Trail.¹⁹⁰ While it may not make sense to allow sidewalk operation on a busy sidewalk, it may make more sense to allow sidewalk operation on

underused facilities in low-density areas next to wide and high-speed arterial roads.

6.3 IS AGE TRULY THE BEST WAY TO REGULATE E-SCOOTERS?

Arguably there is a perception that adolescent e-scooter riders are more likely to be break the rules concerning the operation of e-Scooters.¹⁹¹ For example, Trivedi et al.'s (2019) study found a "significant subset of injuries" among e-scooter riders under the age of 18 (the minimum required age for operating an e-scooter by operators).¹⁹² Additionally, this study found that no adolescent riders (under the age of 25) were using helmets, possibly indicating that some could be violating helmet laws. However, when it comes to speed, this study found adolescent riders appear to be safer riders than adult riders. Regarding age and speed, older riders were found to travel faster than adolescent riders. Importantly, while this finding is more qualitative in nature due to fact the observer determined age groups through observations, it is indicative that age alone may not be the best method to regulate e-scooters due to its varying nature and inconsistency in relation to riding behaviours. There is a conflict present, one where age is not necessarily indicative that e-scooter riders will or will not follow the rules.

Concerning sidewalk versus street riding. one important finding became guite clear. that e-scooter riders will find the path of least resistance. During observations of sidewalk and street e-scooter riders along Santa Clara St., it was noticed that e-scooter riders opted to ride on the sidewalk when the street became far too crowded with vehicular traffic. In particular, there was a section of driveway prior to the street observation zone along Santa Clara St. that allowed for e-scooters to avoid traffic conflicts (such as parked cars in combination with moving vehicles). When e-scooter riders were faced with a restricted or narrowed travel path between parked cars and moving vehicles, they opted instead to jump onto the sidewalk to continue their travels. While this research effort did not specifically measure this phenomenon, it nonetheless proved to be an interesting observational finding, that e-scooter riders when faced with the option will opt to use the sidewalk instead of the street. Of note. there is no bicycle lane or buffer from traffic provided on Santa Clara St. where the current speed limit is 30 mph.¹⁹³ This finding was similar to a study conducted by the City Portland. The study that looked at the relationship between sidewalk riding and street speed limits (where there is a lack of bicycle infrastructure) and found that as street speed limits increased, so did the amount of sidewalk riding.¹⁹⁴ Of

6.4 About Those Pesky Sidewalk RIDERS ON SANTA CLARA ST.

the 128 observed riders in Portland, 50% of riders were found to ride on sidewalks where the posted speed limit was 30 mph while, 18% of riders use the sidewalk where the posted speed limit was 20 mph.¹⁹⁵

6.5 TRAFFIC CONFLICTS ON SIDEWALKS AND MIXED-USE PATHS

Different kinds of traffic conflicts were observed concerning e-scooter riders confronting other road users, including pedestrians, bicyclists, and other e-scooter riders. While traffic conflicts were found among e-scooter riders with those around them, e-scooter riders either slowed down or maneuvered out of the way of the conflict - a positive finding. Additionally, no e-scooter and pedestrian collisions were observed. Temporarily sidestepping the issue associated with sidewalks and narrow space restrictions these two types of reactions exhibit a need for a change in the design of the standard mixed-use path and bicycle lane. Specifically, could any potential traffic conflicts on sidewalks or streets be remedied by simply widening these facilities?

6.6 INCREASING THE RISK OF INJURY? E-SCOOTERS AND VEHICLE COLLISIONS

Two collisions were observed on Santa Clara St. as a result of traffic conflicts between e-scooters and vehicles. No

other forms of collisions were observed between other road users such as pedestrians, cyclists or other e-scooter riders. As a result, it is possible that by restricting e-scooter users to streets as opposed to more buffered facilities, e-scooter riders could be placed at a higher risk injury as a result of collisions with vehicles.

6.7 VOTING WITH AN UNPROTECTED HEAD: E-SCOOTERS AND HELMET USE

With regards to helmets, e-scooter riders rarely wear them, especially when compared to cyclists; however, this could be telling of many different stories. It could verv well be that e-scooter riders are voting with their unprotected heads and view e-scooter riding as not a dangerous activity. Maybe riders need to be better educated about helmet use requirements. although e-scooter applications do inform riders of helmet requirements where they exist. Perhaps carrying around a helmet, for the short portion of a day for a rider is inconvenient and cumbersome. and maybe e-scooter share companies should provide helmets as a result.

This observed phenomenon was also seen in similar micro-mobility sharing schemes such as bike share.¹⁹⁶ For example. Fishman et al. (2013) noted that a survey conducted of Capitol Bike share members revealed 54 percent of users did not wear helmets.¹⁹⁷ This trend continued with Citi Bike in New York

City where 85 percent of users do not wear helmets.¹⁹⁸ Interestingly, Fishman et al. (2013) noted one important trend for low bicycle helmet use: short-term users are less likely to wear helmets when compared with long-term (private bicycle) users who are more likely to wear helmets.¹⁹⁹ Fishman et al.'s finding suggest that since bike share is conducive to more spontaneous trips, users are less likely to have a helmet available.²⁰⁰ Relating the concept of the first and last-mile (which micromobility is meant to solve), the short distance traveled by these devices in combination with spontaneous trips could prove non-conducive to the use of bicycle helmets. Additionally, helmet laws do not encourage the use of bicycle helmets either. In Santa Monica, CA rider compliance with helmet laws is low. estimated to be around approximately two percent.²⁰¹ Similarly, Portland, OR, also has a compliance issue.²⁰²

The results may call into question whether or not regulations should force users to wear helmets. However, it cannot be denied that helmet use is important as 40 percent of all e-scooter injuries are head iniuries.203

6.8 E-Scooter Riders and DISTRACTIONS ARE NOT SO MUCH OF A PROBLEM

E-scooter riders were less distracted

when compared to other modes of micro-mobility, specifically regarding cell phone use. While 16 percent of e-scooter riders wore headphones, it was not possible to determine whether riders were on a call or listening to music. Moreover, in California, it is illegal for bicycle riders to have headphones in both ears.²⁰⁴ As a result, some observed e-scooter riders may be in contravention of the law. While this may pose a hazard to pedestrians and other users around them, cell phone use among e-scooter riders does not seem to pose a threat from a safety perspective. Surprisingly, given worries over distracted travel in other modes. only one rider was trying to use a cellphone. vet unsuccessfully. This may be a byproduct of how a rider operates an e-scooter. Essentially, an e-scooter rider must have both hands on the handlebars as one controls the throttle and the other controls the breaks.

6.9 TALKING WHILE SCOOTING? A POTENTIAL FOR DISTRACTED RIDING

While this research effort can surmise that texting and scootering is not a major issue based on the results, some scooter riders could be distracted by interacting with other scooter riders. De Waard et al.'s (2010) studied noted that a survey among bicyclists involved in collisions revealed that 11.4 percent of riders were talking to other cyclists at the time of the collision.²⁰⁵ As mentioned in Chapter 5, 17 percent of e-scooter riders were observed

riding in packs of two to five people, yet at the same time riders traveling in groups conversed with one another. This number is higher among e-scooter riders when compared to cyclists, indicating that there could be a potential for collisions as a result of group riding.

6.10 REVISITING INFRASTRUCTURE NAMING CONVENTIONS

Observations not considered as part of this study included other emerging modes of micro-motility such as power-assisted unicycles and electric skateboards. These newer emerging modes were observed along with e-scooters. The implication of this finding points to a need to recognize there are many emerging modes of micro-mobility aside from e-scooters flooding city streets. As a result, should cities not be reconsidering existing naming conventions for the bicycle lane? After all, cities like San Jose are actively encouraging through regulations that e-scooters ride on the street, and to use bicycle lanes (where available); thus, does it not make sense to use a more inclusive term reflecting of all modes of micromobility?

6.11 BUT WAIT. THERE'S MORE!

Additional implications were determined based on observations that may have an impact on the design of e-scooters in terms of safety, and an apparent gender imbalance among e-scooter riders.

You!

The current e-scooter design consists of a bell attached to the e-scooter handle bars, but the bells are prone to vandalism and wear and tear. Currently, e-scooters do feature an onboard auditory signal that goes off as a chime to indicate when it is being moved without first being rented. This finding calls into question: how do e-scooter riders signal to others around them, especially on busy streets, of their presence to other road users?

6.11.2 TURNING SIGNALS

Another design issue to consider based on rider behaviour observations I how are e-scooter riders expected to negotiate lane changes and turning without turn signals? While riders could take one hand off the e-scooter to notify drivers behind them, it does prove difficult to balance and operate the vehicle. Thus, riders are either faced with a choice: run the risk of possibly losing balance and falling off the vehicle by removing one of their hands off the handlebars, or do not signal.

After observing 330 e-scooter riders in downtown San Jose, the implications of the findings (based on the results of speed, riding behaviour, and traffic conflicts) showed some inconsistencies with how cities are currently regulating

6.11.1 MOVE OUT OF THE WAY, I CAN'T HEAR

6.12 IMPLICATIONS SUMMARY

e-scooter share. For one, e-scooters riders operate at comparable speeds to that of other modes of micro-mobility. Some of those comparable modes are allowed on the sidewalks in California. while e-scooters are currently restricted from operating on sidewalks (such as San Jose). Additionally, how can cities expect e-scooters (limited to 12 mph in San Jose and 15 mph by the operator) to safely operate in mixed traffic with vehicles that can travel faster than posted speed limits greater than 25 or 35 mph? This conflict has the potential for increasing the risk of injury to e-scooter riders. As mentioned earlier, if riders are provided an opportunity to avoid a highrisk activity such as operating in mixed traffic, they will follow the path of least resistance by riding on the sidewalk. Now, to potential injuries.²⁰⁶ while riders respected the right-of-way for sidewalk users, nonetheless, one has to question whether the incidents of conflicts could be reduced with a simple re-design of existing facilities to accommodate all users.

When it comes to helmet use, it is important to note that some riders might be in contravention of local helmet laws. either due the fact they are underage or simply because the majority do not wear helmets (as shown in this study and by the injury study in Los Angeles). This trend of low compliance is also consistent with other modes of micro-mobility, yet it does call into question whether the age

requirement is the best way to regulate. Simply, there's a conflict concerning the perception that age and alleged safer riding practices are somewhat correlated. despite that fact that in this study, adolescent riders were shown to ride slower than older riders, which one could argue is exhibiting safer riding behaviour.

Concerning e-scooter distractions, riders are less distracted compared to other modes when it comes to cell phone use, yet comparatively distracted when it comes to headphones. Interestingly, there was evidence of talking among group riding observations, which might suggest that e-scootering while talking could be considered a distraction. De Waard et al. noted that this kind of distraction may lead

Finally, with so many emerging modes of micro-mobility that were observed but not part of this study, it does call into question the language cities are using to denote spaces dedicated to micro-mobility. Specifically, does it make sense to call a bicycle lane a "bicycle lane" anymore? Bearing this in mind, the next and final chapter will cover the recommendations for e-scooter share regulations and the design of urban infrastructure as it pertains to speed, rider behaviours, and traffic conflicts.





Based on a comprehensive observational characteristics of a mixed-use path. study of e-scooter riding speed, riding Alternatively, cities may allow e-scooters to ride on sidewalks during off-peak behaviours, and traffic conflicts, along with an understanding of their implications, hours (early in the morning or late at this study makes seven recommendations night) where the sidewalk facility may be for the regulation of e-scooters along wide enough to accommodate multiple with design of urban micro-mobility users. Additionally, cities could consider infrastructure. sidewalk operation in less congested areas such as suburban low-density neighbourhoods with low pedestrian foot 7.1 ALLOW SIDEWALK SCOOTERING

WHERE AND WHEN IT MAKES SENSE USING POSTED SPEED LIMITS

Since e-scooters operate at average speeds at or below 9 mph on sidewalks (well below mixed-trail/path speed limits), it would make sense to regulate the operation of these vehicles where sidewalks mimic the physical

Re-TIMATO FIE STATUS QUO

CHAPTER 7

RETURNED TO THE STATE OF BUILD

7.2 To Avoid Collisions and SIDEWALK RIDING, SEPARATE MICRO-MOBILITY DEVICES FROM VEHICLES

Results showed no incidents of e-scoote riders colliding with other users on streets and sidewalks. One collision, as noted in Chapter 5, was a fall attributed to a

sidewalk curb, the other two could have been easily avoided if e-scooter riders had a safe space that was buffered from parked and vehicular traffic. Specifically, parked cars' side view mirrors that were sticking out into the e-scooter travel path were a cause for concern as two riders hit these mirrors. Pucher and Bhueler's (2016) study of grade separated bicycle infrastructure from traffic noted that buffered and grade-separated facilities can save lives, promote both new and old users to cycle, and reduce the incidence of collisions.²⁰⁷ Specifically, the authors found that in U.S. cities that provided a safe pathway that separated cyclists from high-speed traffic on arterial roads, the number of incidents of collisions dropped from 75 to 25 percent based on 100,000 rides.²⁰⁸ Thus, not only is it necessary

build adequate and safe infrastructure to separate e-scooters from parked and moving traffic, but it is also necessary to add adequate buffered spacing to prevent to the cyclist. collisions with vehicles and potential protruding objects such as side view mirrors. If the City of San Jose expects riders to use Santa Clara St. instead of the sidewalk, then there needs to be adequate and safe facilities in place for riders to use. Otherwise, riders will follow the path of least resistance, and use the adjacent sidewalk.

7.3 BUILDING SLOW. MEDIUM. AND FAST LANES TO ACCOMMODATE ALL TYPES OF RIDERS

In order reduce the number of conflicts with other users. lanes of variable speeds should be implemented in order to accommodate slow-moving traffic, medium-speed moving traffic, and a passing lane for faster-moving traffic on mixed-use paths and existing micromobility (bicycle) infrastructure on streets By giving users the option to travel at their own pace in their own space. the potential for conflicts decrease as faster e-scooter riders can pass others without swerving out of the way in a less predictable fashion when compared to traveling in a straight line. Moreover, this also has the potential to solve any potential conflicts with bicycle riders on the street who were observed traveling at a higher average speed than e-scooter

riders. Even then, only one bicycle conflict was found, and the e-scooter rider in that particular instance slowed down to yield

7.4 INCREASE THE WIDTH OF MICRO-MOBILITY INFRASTRUCTURE TO PROMOTE E-SCOOTER RIDING

E-scooter observations showed group riding on all transportation facilities. While riders were observed chatting with each other, this behaviour could be construed as a distraction but also a tool to encourage more e-scooter riding. This finding implies that since there is evidence of aroup riding, and people want to converse with riders (which promotes group riding), then the design of current urban micro-mobility infrastructure needs to be reconsidered. The solution? Make bicycle or micromobility lanes wider! Currently, the design of bicycle infrastructure allows for riding in single file, but if cities want to encourage more users to utilize various micro-mobility options, then they should make existing micro-mobility infrastructure wider to accommodate more users. When pedestrians walk along the street with friends, family or coworkers, they typically do not walk single file while carrying on a conversation - why would e-scooter riders be any different?

78

7.5 AGE REGULATION: RE-THINKING MANDATORY HELMET LAWS

Current helmet laws do not appear to be encouraging more helmet use as shown among e-scooter riders and similar modes of micro-mobility. Few e-scooter riders are wearing helmets, regardless of age. This is a finding that was similar to that of the e-scooter riders in Los Angeles.

So, what can be done to encourage helmet use in e-scooter share? Helmet design must play a role. Since e-scooter trips are short (less than 3 miles), helmets need to adapt and become convenient for users to travel on short trips.²⁰⁹ Newer helmets on the market are collapsible and can fit into someone's bag. Newer helmet designs could encourage the use of helmets while riding e-scooters.²¹⁰ Additionally, what if cities re-purposed the funds they spent on drafting and enforcing helmet laws, and instead invested in upgrading transportation facilities for all modes of micro-mobility? For example, why not build more grade-separated and barrier-separated micro-mobility lanes on streets?

7.6 FROM THE BICYCLE LANE TO THE "GREEN" LANE

Observations of other emerging micromobility modes were mentioned in Chapter 6 which included power-assisted unicycles and electric skateboards. Cities

need to reconsider the existing naming conventions associated with bicvcle lanes. With more than just e-scooters set to crowd the public right-of-way, not only will the design of the bicycle lane need to change but also a re-branding of the facility itself to better represent the mix of users: a green lane. Some cities currently use the bright green colour to denote bicycle lanes from the rest of the street, yet it is still called a bicycle lane. The bicycle lane perpetuates a perception that only bicycles should be allowed, and is not reflective of the current situation.

This chapter covers research limitations impacted, yet, this method was chosen concerning speed and riding behaviours instead of video recording to reduce the along with future research considerations chance of e-scooter riders altering their on emerging modes of micro-mobility behaviour when known to be observed. and gender imbalances.

8.1 Speed Results

Concerning the speed results, observed riders on the mixed-use path (paseo off of 7th St.) were younger in age due to the proximity of the facility near San Jose State University. This may have resulted in over sampling of similar age ranges. Also, there might have been a potential for human error due to the reaction time required to start and stop the stopwatch as riders crossed the observation zone. As a result, the measured travel time to

ATMATTONS AND FUTURE RESEARCH

CHAPTER 8

CITATIONS AND FURTURE RESEARCH

8.2 RIDING BEHAVIOURS

As it pertains to determining the age of e-scooter riders, there is some subjectivit over the classification of observed riders by age due to the observer determining these age ranges. As a result, there might be a larger proportion of riders represented in the adult category (25 to 50 years of age) that could have fallen either into the younger (adolescent) category of less than years of age, or the age of 50.

8.3 Scooting Forward: Future RESEARCH

Moving forward, there is still much to learn about shared e-scooter programs and e-scooters in general. In terms of operational characteristics, data on the ability of e-scooters to brake and maneuver could be a particularly useful topic for future research. For example, what stopping distance is required for an e-scooter rider on a sidewalk to react to a sudden movement by a nearby pedestrian? Could an e-scooter rider navigate facilities with certain design specifications? Moreover, regarding safety, do e-scooter riders refuse to wear helmets for the same reason as cyclists? Additionally, what would be the operational characteristics of e-scooter

share for riders at night?

8.3.1 WHAT IS COMING DOWN THE PIPELINE?

Cities need to consider all modes of micro-mobility, and not just e-scooters or bicycles. Many different kinds of devices were observed during this study. including:

- Electric skateboards (n=6)
- Electric unicycle (n=1) •
- Skateboards (n=2)
- Roller-blades (in-line skates, n=1)

Electric skateboards were the third highes observed mode (aside from bicycles and e-scooters), and on average traveled at a speed of 15 mph. The presence of alternate modes of micro-mobility should encourage cities to think outside of the box of restricting devices to certain facilities or limiting the maximum operational speed and, instead, focus on developing legislation that is consistent with other modes of micro-mobility that share similar operational and travel behaviour characteristics.

8.3.2 A GENDER IMBALANCE

When comparing the number of apparent females to male riders, results showed that 76 percent of e-scooter riders were male (n=251), while approximately 24 percent were female (n=79). As a result, there is an apparent gender imbalance among e-scooter riders. This issue is not uncommon when it comes to micromobility. A study by Prati (2016) also

found a gender imbalance among cyclists, where the majority of riders tended to be males as opposed to females.²¹¹ Prati (2016) identified two possible reasons for the imbalance among cyclists.²¹² One comes from a study by Askar, Fisher, and Namgung (2013) where the authors noted that safety is a prime consideration for female cyclists regarding their choice to use cycling as a form of transport.²¹³ Moreover, Prati (2016) also stated that in addition to safety, Aldred, Woodstock, and Goodman's (2015) study found that increasing the amount of high-quality cycling infrastructure can lead to higher cycling participation among female riders.²¹⁴ Looking at the results, the gender balance by facility shows that out of 110 street observation (where no bicycle infrastructure was present) 84 percent of riders were male while 16 percent were female. When comparing facilities where pedestrians are present, and grade separated from the roadway, this study found that the percentage of female riders increased: 25 percent for sidewalks, and 32 percent of mixed-use paths. Thus, it prove pertinent to further investigate the gender imbalance among e-scooter riders and see if there is any correlation with other similar modes of micro-mobility.

1. Rory Carroll, "Are Ride-Share Electric Scooters The Future Of Urban Transport?", The Guardian, 2019, accessed April 10, 2019, https://www.theguardian.com/cities/2018/apr/25/ electric-scooters-urban-transport-bird-santa-monica-uk.

2. Julia Sulek, "Watch Out, Bay Area, The E-Scooter Revolution And Adoption Of Electric Scooters In The United (San Is Coming To A Sidewalk Near You", The Mercury News, 2018, accessed April 10, 2019, https://www.mercurynews. com/2018/04/11/watch-out-bay-area-the-e-scooterrevolution-is-coming-to-a-sidewalk-near-you/; Brianne Ebby, "The U.S. Dockless Mobility Experiment", Blog, Eno Transportation Weekly, 2019, accessed April 10, 2019, https://www.enotrans.org/article/the-u-s-dockless-mobilityexperiment/.

3. Brianne Ebby, "The U.S. Dockless Mobility Experiment", Blog, Brianne Ebby, "The U.S. Dockless Mobility Experiment", Blog, Eno Transportation Weekly, 2019, accessed April 10, 2019, https://www.enotrans.org/article/the-u-s-dockless-mobilityexperiment/; Sarah M. Kaufman and Luke Buttenwieser, The State Of Scooter Sharing In United States Cities (New York: Rudin Center for Transportation, 2018), accessed April 10, 2019, https://wagner.nyu.edu/files/faculty/publications/Rudin ScooterShare Aug2018 0.pdf.

4. Alissa Walker, "Don't Ban Scooters. Redesign Streets.", Blog, Curbed: San Francisco, 2019, accessed April 10, 2019, https://www.curbed.com/word-on-thestreet/2018/7/13/17246060/scooters-uber-lyft-bird-limestreets.

5. Populus, The Micro-Mobility Revolution: The Introduction And Adoption Of Electric Scooters In The United (San Francisco: Populus Al, 2018), accessed April 10, 2019, http:// research.populus.ai/reports/ Populus MicroMobility 2018-July. pdf.

6.lbid.

7.Rory Carroll, "Are Ride-Share Electric Scooters The Future Of Urban Transport?", The Guardian, 2019, accessed April 10, 2019, https://www.theguardian.com/cities/2018/apr/25/ electric-scooters-urban-transport-bird-santa-monica-uk.

8. Elliot Fishman. "Bikeshare: A Review Of Recent Literature". Transport Reviews 36, no. 1 (2015): 92-113, accessed April 10, 2019, http://10.1080/01441647.2015.1033036.

9. Populus, The Micro-Mobility Revolution: The Introduction Francisco: Populus Al, 2018), accessed April 10, 2019, http:// research.populus.ai/reports/ Populus MicroMobility 2018-July. pdf.

10. Julia Sulek, "Watch Out, Bay Area, The E-Scooter Revolution Is Coming To A Sidewalk Near You", The Mercury News, 2018, accessed April 10, 2019, https://www. mercurynews.com/2018/04/11/watch-out-bay-area-thee-scooter-revolution-is-coming-to-a-sidewalk-near-you/; Eno Transportation Weekly, 2019, accessed April 10, 2019, https://www.enotrans.org/article/the-u-s-dockless-mobilityexperiment/.

11."Bye Bye, Birdie: Beverly Hills Bans Electric Scooters For 6 Months, Pending Regulation", Losangeles.Cbslocal.Com, last modified 2018, accessed April 10, 2019, https://losangeles. cbslocal.com/2018/07/24/beverly-hills-bans-bird-limeelectric-scooters/.

12 Tarak K. Trivedi et al., "Injuries Associated With Standing Electric Scooter Use", JAMA Network Open 2, no. 1 (2019): e187381, accessed April 10, 2019, http://:10.1001/ iamanetworkopen.2018.7381.

13. Jaileane Aguilar, "Motorized Scooters Restricted", Spartan Daily, 2019, accessed April 10, 2019, http://www.sisunews. com/spartan daily/news/article 0cc85350-3486-11e9-968c-3f72417da000.html.

14. Aarian Marshall, "As Shared Scooters Invade, Cities Must Decide Who's Vehicles Belong Where", Blog, Wired, 2018, accessed April 10, 2019, https://www.wired.com/story/sanfrancisco-electric-scooters-bird-limebike/.

15. Julia Sulek, "Watch Out, Bay Area, The E-Scooter Revolution Is Coming To A Sidewalk Near You", The

Mercury News, 2018, accessed April 10, 2019, https://www. mercurynews.com/2018/04/11/watch-out-bay-area-the-escooter-revolution-is-coming-to-a-sidewalk-near-you/.

16. "Dockless Mobility Vehicle Pilot Permit Program Overview", Denvergov.Org. last modified 2018, accessed April 10, 2019, https://www.denvergov.org/content/dam/denvergov/ Portals/705/documents/permits/Dockless-Mobility-Pilot-Permit-Program-Overview June2018.pdf.

17. Aarian Marshall, "As Shared Scooters Invade, Cities Must Decide Who's Vehicles Belong Where", Blog, Wired, 2018, accessed April 10, 2019, https://www.wired.com/story/sanfrancisco-electric-scooters-bird-limebike/; Elena Weissmann, "Enjoy Bird Scooters While You Can, Because They Likely Won't Be Back Until 2019", Milwaukee Journal Sentinel, 2019, accessed April 10, 2019, https://www.jsonline.com/ story/news/local/milwaukee/2018/07/20/facing-ban-birdscooters-may-disappear-milwaukee-until-2019/803014002/.

18. Ben Jose, "SFMTA Offers Two Permits For One-Year Powered Scooter Pilot", Blog, SFMTA, 2018, accessed April 10, 2019, https://www.sfmta.com/blog/sfmta-offers-twopermits-one-year-powered-scooter-pilot.

19. Brianne Ebby, "The U.S. Dockless Mobility Experiment". Blog, Eno Transportation Weekly, 2019, accessed April 10, 2019, https://www.enotrans.org/article/the-u-sdockless-mobility-experiment/; Sarah M. Kaufman and Luke Buttenwieser, The State Of Scooter Sharing In United States Cities (New York: Rudin Center for Transportation, 2018). accessed April 10, 2019, https://wagner.nyu.edu/files/faculty/ publications/Rudin ScooterShare Aug2018 0.pdf.

20. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

21. lbid.

22. San Francisco Municipal Transportation Agency (SFMTA), Pilot Scooter Share Program Permit Staff Report (San Francisco: San Francisco Municipal Transportation Agency (SFMTA), 2018), accessed April 10, 2019, https://www.sfmta.com/ sites/default/files/reports-and-documents/2018/04/5-1-18 30. Ibid. item 11 pilot scooter share program permit.docx .pdf.

23. Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the Transportation Research Board 1773, no. 1 (2001); 47-55. accessed April 10, 2019, https://doi.org/10.3141/1773-06; Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010; Bruce W. Landis et al., "Characteristics Of Emerging Road And Trail Users And Their Safety". Transportation Research Record: Journal of the Transportation Research Board 1878, no. 1 (2004): 131-139. accessed April 10, 2019, https://doi.org/10.3141/1878-16; Sheryl Miller et al., "Segway Rider Behavior", Transportation Research Record: Journal of the Transportation Research Board 2073, no. 1 (2008): 125-132, accessed April 10, 2019, https://doi.org/10.3141/2073-14.

24. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

25. lbid.

26. Populus. The Micro-Mobility Revolution: The Introduction And Adoption Of Electric Scooters In The United (San Francisco: Populus Al, 2018), accessed April 10, 2019, http:// research.populus.ai/reports/ Populus MicroMobility 2018-July. pdf.

27. Brian Alberts, Jamie Palumbo and Eric Pierce, Vehicle 4 Change: Health Implications Of The Capital Bikeshare Program (Washington D.C.: The George Washington University, Master of Public Policy and Public Administration Program, 2013). accessed April 28, 2019, http://mobilitylab.org/wpcontent/ uploads/2013/05/2013 CapitalBikeshare HealthSurvey Report.pdf.

28. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

29. Michael Gagliano, Buffalo Bike Share Demonstration Project, Report (Buffalo, New York: Shared Mobility Inc., 2015), accessed April 28, 2019, https://www.nyserda.ny.gov.

31. lbid.

32. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf; Populus, The Micro-Mobility Revolution: The Introduction And Adoption Of Electric Scooters In The United (San Francisco: Populus Al, 2018). accessed April 10, 2019, http://research.populus.ai/reports/ Populus MicroMobility 2018-July.pdf.

33. Ibid.

34. lbid.

35. Populus, The Micro-Mobility Revolution: The Introduction And Adoption Of Electric Scooters In The United (San Francisco: Populus Al, 2018), accessed April 10, 2019, http:// pdf. research.populus.ai/reports/ Populus MicroMobility_2018-July. pdf.

36. Sarah M. Kaufman and Luke Buttenwieser, The State Of Scooter Sharing In United States Cities (New York: Rudin Center for Transportation, 2018), accessed April 10, 2019, https://wagner.nyu.edu/files/faculty/publications/Rudin ScooterShare Aug2018 0.pdf.

37. Jonathan Shieber, "Bird Buys More Scooters", Blog, Tech Crunch, 2018, accessed April 28, 2019, https://techcrunch. com/2018/05/11/in-a-bid-to-corner-supply-bird-locks-inexclusive-deals-with-the-biggest-scooter-vendors; Meagan R. Dickey, "Lime Partners With Segway To Build Electric Scooters", Blog, Tech Crunch, 2018, accessed April 28, 2019, https://techcrunch.com/2018/05/11/in-a-bid-to-cornersupply-bird-locks-in-exclusive-deals-with-the-biggestscooter-vendors.

38. Ibid.

39. Ibid.

40. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf; Populus, The Micro-Mobility Revolution: The Introduction And Adoption Of Electric Scooters

In The United (San Francisco: Populus AI, 2018), accessed April 10, 2019, http://research.populus.ai/reports/ Populus MicroMobility 2018–July.pdf; Sarah M. Kaufman and Luke Buttenwieser, The State Of Scooter Sharing In United States Cities (New York: Rudin Center for Transportation, 2018). accessed April 10, 2019, https://wagner.nyu.edu/files/faculty/ publications/Rudin ScooterShare Aug2018 0.pdf.

41. Populus. The Micro-Mobility Revolution: The Introduction And Adoption Of Electric Scooters In The United (San Francisco: Populus Al, 2018), accessed April 10, 2019, http:// research.populus.ai/reports/ Populus MicroMobility 2018-July. pdf.

42. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

43. Populus. The Micro-Mobility Revolution: The Introduction And Adoption Of Electric Scooters In The United (San Francisco: Populus Al, 2018), accessed April 10, 2019, http:// research.populus.ai/reports/ Populus MicroMobility 2018-July.

44. Sean Keenen. "Bird Introduces Scooter Delivery Service". Blog, Curbed: Atlanta, 2018, accessed April 28, 2019, https:// atlanta.curbed.com/2018/10/9/17953718/bird-scooterdelivery-service-last-mile-shareable-vehicles.

45. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

46. lbid.

47. lbid.

48. Adam Brinklow, "Electric Scooter Companies Discount Trips For Students, Low-Income Riders", Blog, Curbed; San Francisco, 2018, accessed April 28, 2019, https://sf.curbed. com/2018/7/20/17596320/scooters-san-francisco-limebird-discounts-permits-students.

49. Alexis C. Madrigal. "San Francisco's Scooter War Is Over". Blog, Citylab, 2018, accessed April 28, 2019, https://www. citylab.com/transportation/2018/10/san-franciscos-scooterwar-is-over-and-the-scooters-won/573022/.

50. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation

And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

51. Aarian Marshall, "Skip Scooters Get A Latch So They Don't Junk Up The Sidewalks", Blog, Wired Magazine, 2018, accessed April 28, 2019, https://www.wired.com/story/skipscooters-latch-dont-iunk-up-sidewalk/.

52. Ibid.

53. Melia Robinson, "A Startup In The West Coast Scooter Sharing Craze Is Already Worth \$1 Billion - Here's What It's Like To Ride A Bird Scooter". Blog. Business Insider, 2018. accessed April 28, 2019, https://www.businessinsider.com/ bird-electric-scooter-review-2018-4.

54. Ibid.

55. Ibid.

56. Melia Robinson, "A Startup In The West Coast Scooter Sharing Craze Is Already Worth \$1 Billion - Here's What It's Like To Ride A Bird Scooter", Blog, Business Insider, 2018, accessed April 28, 2019, https://www.businessinsider.com/ bird-electric-scooter-review-2018-4.

57. Ibid.

58. lbid.

59. Melia Robinson, "A Startup In The West Coast Scooter Sharing Craze Is Already Worth \$1 Billion - Here's What It's Like To Ride A Bird Scooter", Blog, Business Insider, 2018, accessed April 28, 2019, https://www.businessinsider.com/ bird-electric-scooter-review-2018-4.

- 60. lbid.
- 61. lbid.
- 62. lbid.

63. Melia Robinson. "A Startup In The West Coast Scooter Sharing Craze Is Already Worth \$1 Billion - Here's What It's Like To Ride A Bird Scooter", Blog, Business Insider, 2018, accessed April 28, 2019, https://www.businessinsider.com/ bird-electric-scooter-review-2018-4.

64. Chris Bank. "The Rider Experience Of Micro-Mobility Products: Scooters, Bikes & Beyond", Blog, UX Collective, 2018, accessed April 28, 2019, https://uxdesign.cc/a-usersexperience-with-micro-mobility-5700c5c5104c.

65. Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the

Transportation Research Board 1773, no. 1 (2001): 47-55, accessed April 10, 2019, https://doi.org/10.3141/1773-06; Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010; Bruce W. Landis et al., "Characteristics Of Emerging Road And Trail Users And Their Safety". Transportation Research Record: Journal of the Transportation Research Board 1878, no. 1 (2004): 131-139. accessed April 10, 2019, https://doi.org/10.3141/1878-16; Sheryl Miller et al., "Segway Rider Behavior", Transportation Research Record: Journal of the Transportation Research Board 2073, no. 1 (2008): 125-132, accessed April 10, 2019. https://doi.org/10.3141/2073-14.

66. Melanie Curry, "CA Bill No Longer Redefines E-Scooters, But Removes Helmet Requirements For Over-18". Blog. Streetsblog: California, 2018, accessed April 28, 2019, https:// cal.streetsblog.org/2018/07/03/ca-bill-no-longer-redefinese-scooters-but-removes-helmet-requirements-for-over-18; Adam Brinklow, "Electric Scooter Companies Discount Trips For Students, Low-Income Riders", Blog, Curbed: San Francisco, 2018, accessed April 28, 2019, https://sf.curbed. com/2018/7/20/17596320/scooters-san-francisco-limebird-discounts-permits-students.

67. Melanie Curry, "CA Bill No Longer Redefines E-Scooters, But Removes Helmet Requirements For Over-18". Blog. Streetsblog: California, 2018, accessed April 28, 2019, https:// cal.streetsblog.org/2018/07/03/ca-bill-no-longer-redefinese-scooters-but-removes-helmet-requirements-for-over-18; Adam Brinklow, "Electric Scooter Companies Discount Trips For Students, Low-Income Riders", Blog, Curbed: San Francisco, 2018, accessed April 28, 2019, https://sf.curbed. com/2018/7/20/17596320/scooters-san-francisco-limebird-discounts-permits-students.

68 Ibid

69. Melissa Etehad and Tracey Lien, "Santa Monica Tells Scooter Companies: You Can Stay - But There Are New Rules", The Los Angeles Times, 2018, accessed April 28, 2019. https://www.latimes.com/business/technology/la-fitn-santa-monica-scooter-rules-20180613-story.html: Laura J. Nelson, "L.A. Approves Rules For Thousands Of Scooters, With A 15-Mph Speed Limit And Aid For Low-Income Riders", The Los Angeles Times, 2018, accessed April 28, 2019, https://www.latimes.com/local/lanow/la-me-ln-scootervote-20180904-story.html: Fredrick Kunkle, "Scooters Set To Return To San Francisco", The Washington Post, 2018, accessed April 28, 2019, https://www.washingtonpost. com/transportation/2018/10/14/scooters-set-return-sanfrancisco/?noredirect=on&utm term=.43ac8deb7bd9.

70. Robin Abacarion. "Cities Crack Down On Electric Scooter

Scofflaws, While Trying To Untangle The Chaos Of A New Technology", Los Angeles Times, 2018, accessed November 9, 2018, http://www.latimes.com/la-me-abcarian-scooters-2018072-story.html.

71. Alison Griswold, "A US City Thinks Electric Scooters Are "Toys" And Should Be Ridden On The Sidewalk", Blog, Quartz, 2018. accessed April 28. 2019. https://gz.com/1386718/ denver-savs-electric-scooters-are-tovs-and-should-beridden-on-the-sidewalk/.

72. lbid.

73. Emily Deruy, "San Jose Imposes New Scooter Regulations", The Mercury News, 2018, accessed April 28, 2019, https:// www.mercurynews.com/2018/12/19/san-jose-imposesnew-scooter-regulations; Jaileane Aguilar, "Motorized Scooters Restricted". Spartan Daily, 2019, accessed April 10. 2019. http://www.sisunews.com/spartan_daily/news/ article 0cc85350-3486-11e9-968c-3f72417da000.html.

74. Elijah Chiland, "California Removes Helmet Requirement For Electric Scooters", Blog, Curbed: Los Angeles, 2018, accessed April 28, 2019, https://la.curbed.com/2018/9/21/17884220/ bird-lime-scooters-rules-helmets-california.

75. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

76. Sarah M. Kaufman and Luke Buttenwieser, The State Of Scooter Sharing In United States Cities (New York: Rudin Center for Transportation, 2018), accessed April 10, 2019,https://wagner.nyu.edu/files/faculty/publications/Rudin Scooter Share Aug2018 0.pdf.

77. Laura J. Nelson, "L.A. Approves Rules For Thousands Of Scooters, With A 15-Mph Speed Limit And Aid For Low-Income Riders", The Los Angeles Times, 2018, accessed April 28, 2019, https://www.latimes.com/local/lanow/la-me-lnscooter-vote-20180904-story.html.

78. "San Jose, CA - Official Website - Micro Mobility Vendor Page - Shared Micro-Mobility Permit Administrative Regulations", Sanjoseca.Gov, last modified 2018, accessed April 28, 2019, http://www.sanioseca.gov/index. aspx?NID=6303.

79. "Policies + Laws • Peopleforbikes", Peopleforbikes, accessed April 28, 2019, https://peopleforbikes.org/ourwork/e-bikes/policies-and-laws/.

80. Aarian Marshall, "SF Is Bringing Back Electric Scooters With

Limits", Blog, WIRED, 2018, accessed April 28, 2019, https:// www.wired.com/story/san-francisco-santa-monica-e-scooterpermits/.

81. Alissa Walker, "Bird, Lime, Lyft, Uber Will All Be Allowed To Operate Scooters In Santa Monica", Blog, Curbed: Los Angeles, 2018, accessed April 28, 2019, https://la.curbed. com/2018/8/30/17800498/scooters-santa-monica-birdlime-lvft-uber.

82. National Association of City Transportation Officials (NACTO), NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1 (New York, 2018), accessed April 10, 2019, https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

83. Elena Weissmann, "Enjoy Bird Scooters While You Can, Because They Likely Won't Be Back Until 2019". Milwaukee Journal Sentinel, 2019, accessed April 10, 2019, https://www. jsonline.com/story/news/local/milwaukee/2018/07/20/ facing-ban-bird-scooters-may-disappear-milwaukeeuntil-2019/803014002/; Alissa Walker, "Everything You Need To Know About Scooters, Bike Share, Dockless Bikes In Los Angeles", Blog, Curbed: Los Angeles, 2018, accessed April 28, 2019, https://la.curbed.com/2018/7/31/17623336/birdlime-scooter-bike-app-locations-cost-hours; Robin Abacarion, 93. Ibid. "Cities Crack Down On Electric Scooter Scofflaws, While Trving To Untangle The Chaos Of A New Technology", The Los Angeles Times, 2018, accessed April 28, 2019, http://www. latimes.com/la-me-abcarian-scooters-2018072-story.html.

84. Alissa Walker, "Everything You Need To Know About Scooters, Bike Share, Dockless Bikes In Los Angeles", Blog, Curbed: Los Angeles, 2018, accessed April 28, 2019, https:// la.curbed.com/2018/7/31/17623336/bird-lime-scooterbike-app-locations-cost-hours.

85. Melanie Curry, "CA Bill No Longer Redefines E-Scooters, But Removes Helmet Requirements For Over-18". Blog. Streetsblog: California, 2018, accessed April 28, 2019, https:// cal.streetsblog.org/2018/07/03/ca-bill-no-longer-redefinese-scooters-but-removes-helmet-requirements-for-over-18/; Adam Brinklow, "Electric Scooter Companies Discount Trips For Students, Low-Income Riders", Blog, Curbed; San Francisco, 2018, accessed April 28, 2019, https://sf.curbed. com/2018/7/20/17596320/scooters-san-francisco-limebird-discounts-permits-students.

86. Lizzy Acker. "I Took An Electric Scooter On A 12.5-Mile Journey Through Portland -- Here's What Happened", Blog, The Oregonian: Live, 2018, accessed April 28, 2019, https:// www.oregonlive.com/portland/2018/07/i took an electric scooter on.html.

87. Ibid.

SUIDAT DAIXERS, OR TOO FAST AND TOO FURIOUS?

88. Melissa Etehad and Tracey Lien, "Santa Monica Tells Scooter Companies: You Can Stay — But There Are New Rules", The Los Angeles Times, 2018, accessed April 28, 2019, https://www.latimes.com/business/technology/la-fi-tnsanta-monica-scooter-rules-20180613-story.html.

89. Lizzy Acker, "I Took An Electric Scooter On A 12.5-Mile Journey Through Portland -- Here's What Happened". Blog. The Oregonian: Live, 2018, accessed April 28, 2019, https:// www.oregonlive.com/portland/2018/07/i took an electric scooter on.html.

90. David Gutman, "Helmets May Be Seattle Law, But Many Bike-Share Riders Don'T Wear Them". The Seattle Times. 2017. accessed April 28. 2019. https://www.seattletimes.com/ seattle-news/transportation/helmets-may-be-seattle-law-butmany-bike-share-riders-dont-wear-them/.

91. Federal Highway Administration (FHWA). Characteristics Of Emerging Road And Trail Users And Their Safety, FHWA-HRT-04-104 (Federal Highway Administration, 2004), accessed April 28, 2019, https://www.fhwa.dot.gov/ publications/research/safety /04104/.

92. lbid.

94. "Article 6. Electric Personal Assistive Mobility Devices [21280 - 21282]", Leginfo.Legislature.Ca.Gov, last modified 2002, accessed April 28, 2019, http://leginfo. legislature.ca.gov/faces/codes displayText.xhtml?lawCode =VEH&division=11.&title=&part =&chapter=1.&article=6.

95. Richard L. Knoblauch, Martin T. Pietrucha and Marsha Nitzburg, "Field Studies Of Pedestrian Walking Speed And Start-Up Time", Transportation Research Record: Journal of the Transportation Research Board 1538, no. 1 (1996): 27-38, accessed April 28, 2019, https://doi.org/10.1177/036119819 Safety Performance Of Skateboard Travel: A Look At Injury 6153800104.

96. Ibid

97. Federal Highway Administration (FHWA). Characteristics Of Emerging Road And Trail Users And Their Safety, FHWA-HRT-04-104 (Federal Highway Administration, 2004), accessed April 28, 2019, https://www.fhwa.dot.gov/ publications/research/safety /04104/.

98. Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters", Transportation Research Record: Journal of the Transportation Research Board 1773, no. 1 (2001): 47-55. accessed April 10, 2019, https://doi.org/10.3141/1773-06.

99. Federal Highway Administration (FHWA), Characteristics Of Emerging Road And Trail Users And Their Safety, FHWA- HRT-04-104 (Federal Highway Administration, 2004), accessed April 28, 2019, https://www.fhwa.dot.gov/ publications/research/safety/04104/.

100. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.ora/10.1016/i.ith.2017.08.010.

101. Jennifer Dill and Geoffrey Rose, "Electric Bikes And Transportation Policy", Transportation Research Record: Journal of the Transportation Research Board 2314, no. 1 (2012); 1-6. accessed April 28, 2019, https://doi.org/10.3141/2314-01,

102. Federal Highway Administration (FHWA), Characteristics Of Emerging Road And Trail Users And Their Safety, FHWA-HRT-04-104 (Federal Highway Administration, 2004). accessed April 28, 2019, https://www.fhwa.dot.gov/ publications/research/safety /04104/.

103. K. K. Finnis and D. Walton, "Field Observations To Determine The Influence Of Population Size, Location And Individual Factors On Pedestrian Walking Speeds", Ergonomics 51, no. 6 (2008); 827-842, accessed April 28, 2019, https://doi.org/10.1080/00140130701812147; Dick de Waard et al., "Mobile Phone Use While Cycling: Incidence And Effects On Behaviour And Safety", Ergonomics 53, no. 1 (2010): 30-42, accessed April 28, 2019, https://doi. org/10.1080/00140130903381180.

104. K. K. Finnis and D. Walton, "Field Observations To Determine The Influence Of Population Size, Location And Individual Factors On Pedestrian Walking Speeds", Ergonomics 51, no. 6 (2008): 827-842, accessed April 28, 2019, https:// doi.org/ 10.1080/00140130701812147.

105. Kevin Fang and Susan Handy, "Skate And Die? The Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010; Richard L. Knoblauch, Martin T. Pietrucha and Marsha Nitzburg, "Field Studies Of Pedestrian Walking Speed And Start-Up Time". Transportation Research Record: Journal of the Transportation Research Board 1538, no. 1 (1996): 27-38, accessed April 28, 2019, https://doi.org/10.1177/0361198196153800104.

106. Joseph Broach, Jennifer Dill and John Gliebe, "Where Do Cyclists Ride? A Route Choice Model Developed With Revealed Preference GPS Data", Transportation Research Part A: Policy and Practice 46, no. 10 (2012): 1730-1740. accessed April 28, 2019. https://doi.org/10.1016/i. tra.2012.07.005; Lisa Aultman-Hall and Jeffrev LaMondia. "Evaluating The Safety Of Shared-Use Paths", Transportation Research Record: Journal of the Transportation Research

Board 1939, no. 1 (2005): 99-106, accessed April 28, 2019, https://doi.org/10.1177/0361198105193900112. ; Jonathan X. Weinert et al., "Electric Two-Wheelers In China", Transportation Research Record: Journal of the Transportation Research Board 2038, no. 1 (2007); 62-68, accessed April 28, 2019, https://doi.org/10.3141/2038-08; Yi Ai, Zongping Li and Mi Gan, "A Solution To Measure Traveler's Transfer Tolerance For Walking Mode And Dockless Bike-Sharing Mode". The Journal of Supercomputing (2018), accessed April 10, 2019, https://doi.org/10.3141/2073-14. 28, 2019, https://doi.org/10.1007/s11227-017-2211-7.

107. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010; Yi Ai, Zongping Li and Mi Gan, "A Solution To Measure Traveler's Transfer Tolerance For And Trail Users And Their Safety", Transportation Research Walking Mode And Dockless Bike-Sharing Mode". The Journal of Supercomputing (2018), accessed April 28, 2019, https:// doi.org/10.1007/s11227-017-2211-7.

108. Jonathan X. Weinert et al., "Electric Two-Wheelers In China". Transportation Research Record: Journal of the Transportation Research Board 2038, no. 1 (2007); 62-68. accessed April 28, 2019, https://doi.org/10.3141/2038-08,

109. Lisa Aultman-Hall and Jeffrey LaMondia, "Evaluating The Safety Of Shared-Use Paths". Transportation Research Record: Journal of the Transportation Research Board 1939, no. 1 (2005): 99-106, accessed April 28, 2019, https://doi.org/10.1 Of Electric Bicycles And Bicycles", Transportation Research 177/0361198105193900112.

110. Yi Ai, Zongping Li and Mi Gan, "A Solution To Measure Traveler's Transfer Tolerance For Walking Mode And Dockless Bike-Sharing Mode", The Journal of Supercomputing (2018), accessed April 28, 2019, https://doi.org/10.1007/s11227-017-2211-7.

111. Bruce W. Landis et al., "Characteristics Of Emerging Road And Trail Users And Their Safety". Transportation Research Record: Journal of the Transportation Research Board 1878, no. 1 (2004): 131-139, accessed April 10, 2019, https://doi. org/10.3141/1878-16; Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the Transportation Research Board 1773. no. 1 (2001): 47-55, accessed April 10, 2019, https://doi. org/10.3141/1773-06.

112. Bruce Landis et al., "Characteristics Of Emerging Road And Trail Users And Their Safety". Transportation Research Record: Journal of the Transportation Research Board 1878 (2004): 131-139. accessed October 26, 2018, https://doi. org/10.3141/1878-16; Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the Transportation Research Board 1773 (2001): 47-55, accessed October 26, 2018, https://doi.

org/10.3141/1773-06; Sheryl Miller et al., "Segway Rider Behavior", Transportation Research Record: Journal of the Transportation Research Board 2073, no. 1 (2008): 125-132, accessed April 10, 2019, https://doi.org/10.3141/2073-1.

113. Sheryl Miller et al., "Segway Rider Behavior", Transportation Research Record: Journal of the Transportation Research Board 2073, no. 1 (2008): 125-132, accessed April

114. Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the Transportation Research Board 1773, no. 1 (2001): 47-55. accessed April 10, 2019, https://doi.org/10.3141/1773-06,

115. Bruce W. Landis et al., "Characteristics Of Emerging Road Record: Journal of the Transportation Research Board 1878. no. 1 (2004); 131–139, accessed April 10, 2019, https://doi. ora/10.3141/1878-16.

116. lbid.

117. Sheryl Miller et al., "Segway Rider Behavior", Transportation Research Record: Journal of the Transportation Research Board 2073, no. 1 (2008): 125-132, accessed April 10, 2019, https://doi.org/10.3141/2073-14.

118. Sen Lin et al., "Comparison Study On Operating Speeds Record: Journal of the Transportation Research Board 2048. no. 1 (2008): 52-59, accessed April 28, 2019, https://doi. ora/10.3141/2048-07.

119. Sheryl Miller et al., "Segway Rider Behavior", Transportation Research Record: Journal of the Transportation Research Board 2073, no. 1 (2008): 125-132, accessed April 10, 2019, https://doi.org/10.3141/2073-14.

120. Ibid.

121. Ibid.

122. Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the Transportation Research Board 1773, no. 1 (2001): 47-55, accessed April 10, 2019, https://doi.org/10.3141/1773-06. 123. Ibid.

124. Kevin Fang and Susan Handy. "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016 /j.jth.2017.08.010.

90

125. Ibid.

126. Ibid. 127. Ibid.

128. Ibid.

129. Dick de Waard et al., "Mobile Phone Use While Cycling: Incidence And Effects On Behaviour And Safety". Ergonomics 53. no. 1 (2010); 30-42. accessed April 28. 2019. https://doi. org/10.1080/00140130903381180.

130. Ibid.

131. Ibid.

132. Dick de Waard et al., "Mobile Phone Use While Cycling: Incidence And Effects On Behaviour And Safety", Ergonomics 53. no. 1 (2010); 30-42. accessed April 28. 2019. https://doi. org/10.1080/00140130903381180.

133. Ibid.

134. Richard L. Knoblauch, Martin T. Pietrucha and Marsha Nitzburg, "Field Studies Of Pedestrian Walking Speed And Start-Up Time". Transportation Research Record: Journal of the Transportation Research Board 1538, no. 1 (1996): 27-38, accessed April 28, 2019, https://doi.org/10.1177/036119819 6153800104

135. K. K. Finnis and D. Walton, "Field Observations To Determine The Influence Of Population Size, Location And Individual Factors On Pedestrian Walking Speeds", Ergonomics 51, no. 6 (2008): 827-842, accessed April 28, 2019, https:// doi.org/10.1080/00140130701812147.

136. Ibid.

137. Ibid.

138. Sen Lin et al., "Comparison Study On Operating Speeds Of Electric Bicycles And Bicycles", Transportation Research Record: Journal of the Transportation Research Board 2048, no. 1 (2008): 52-59, accessed April 28, 2019, https://doi. ora/10.3141/2048-07.

139. lbid.

140. Melissa Etehad and Tracev Lien. "Santa Monica Tells Scooter Companies: You Can Stav — But There Are New Rules". The Los Angeles Times. 2018. accessed April 28. 2019, https://www.latimes.com/business/technology/la-fitn-santa-monica-scooter-rules-20180613-story.html; Laura J. Nelson, "L.A. Approves Rules For Thousands Of Scooters. With A 15-Mph Speed Limit And Aid For Low-Income Riders", The Los Angeles Times, 2018, accessed April 28, 2019, https://www.latimes.com/local/lanow/la-me-ln-scooterthe-sidewalk/.

scooter on.html.

142. Melanie Curry, "CA Bill No Longer Redefines E-Scooters, But Removes Helmet Requirements For Over-18". Blog. Streetsblog: California, 2018, accessed April 28, 2019, https:// cal.streetsblog.org/2018/07/03/ca-bill-no-longer-redefinese-scooters-but-removes-helmet-requirements-for-over-18/.

144. Melanie Curry, "CA Bill No Longer Redefines E-Scooters, But Removes Helmet Requirements For Over-18", Blog, Streetsblog: California, 2018, accessed April 28, 2019, https:// cal.streetsblog.org/2018/07/03/ca-bill-no-longer-redefinese-scooters-but-removes-helmet-requirements-for-over-18/: Adam Brinklow, "Electric Scooter Companies Discount Trips For Students, Low-Income Riders", Blog, Curbed: San Francisco, 2018, accessed April 28, 2019, https://sf.curbed. com/2018/7/20/17596320/scooters-san-francisco-limebird-discounts-permits-students.

145. Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters", Transportation Research Record: Journal of the Transportation Research Board 1773, no. 1 (2001): 47-55. accessed April 10, 2019, https://doi.org/10.3141/1773-06,

146. Shawn M. Turner et al., Travel Time Collection Handbook (Washington D.C.: Federal Highway Administration, 1998), accessed April 29, 2019, https://www.fhwa.dot.gov/ohim/ tvtw/natmec/00020.pdf.

147. Federal Highway Administration (FHWA), Characteristics

ETOTOTAS

vote-20180904-story.html.; Fredrick Kunkle, "Scooters Set To Return To San Francisco", The Washington Post, 2018, accessed April 28, 2019, https://www.washingtonpost. com/transportation/2018/10/14/scooters-set-return-sanfrancisco/?noredirect=on &utm term=.43ac8deb7bd9: Alison Griswold, "A US City Thinks Electric Scooters Are "Toys" And Should Be Ridden On The Sidewalk", Blog, Quartz, 2018, accessed April 28, 2019, https://gz.com/1386718/denversavs-electric-scooters-are-tovs-and-should-be-ridden-on-

141. Lizzy Acker, "I Took An Electric Scooter On A 12.5-Mile Journey Through Portland -- Here's What Happened", Blog, The Oregonian: Live, 2018, accessed April 28, 2019, https:// www.oregonlive.com/portland/2018/07/i took an electric

143. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010; Richard L. Knoblauch, Martin T. Pietrucha and Marsha Nitzburg, "Field Studies Of Pedestrian Walking Speed And Start-Up Time", Transportation Research Record: Journal of the Transportation Research Board 1538, no. 1 (1996): 27-38, accessed April 28, 2019, https://doi.org/10.1177/0361198196153800104.

Of Emerging Road And Trail Users And Their Safety, FHWA-HRT-04-104 (Federal Highway Administration, 2004), accessed April 28, 2019, https://www.fhwa.dot.gov/ publications/research/safety/04104/; Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the Transportation Research Board 1773, no. 1 (2001): 47-55, accessed April 10, 2019. https://doi.org/10.3141/1773-06: Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https://doi.org/10.1016/j ith.2017.08.010.

148. "Detail", Dmv.Ca.Gov, last modified 2018, accessed April 29, 2019, https://www.dmv.ca.gov/portal/dmv/detail/vr/ scooters.

149. Tarak K. Trivedi et al., "Injuries Associated With Standing Electric Scooter Use", JAMA Network Open 2, no. 1 (2019): e187381, accessed April 10, 2019, http://:10.1001/ jamanetworkopen.2018.7381.

150. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010.

151. Ibid.

152. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010.

153. Ibid.

154. Bicycle And Pedestrian Safety Brochure, ebook (Santa Clara County: Office of Traffic Safety of California, 2019), accessed April 29, 2019, https://www.ots.ca.gov/wp-content/ uploads/sites/67/2018/09/BicyclePedSafetyBrochure.pdf.

155. Dick de Waard et al., "Mobile Phone Use While Cycling: Incidence And Effects On Behaviour And Safety", Ergonomics 53, no. 1 (2010): 30-42, accessed April 28, 2019, https://doi. ora/10.1080/00140130903381180.

156. Kevin Fang and Susan Handy. "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010.

157. Federal Highway Administration (FHWA), Characteristics

Of Emerging Road And Trail Users And Their Safety, FHWA-HRT-04-104 (Federal Highway Administration, 2004), accessed April 28, 2019, https://www.fhwa.dot.gov/ publications/research/safety/04104/.

158. "San Jose, CA - Official Website - Micro Mobility Vendor Page - Shared Micro-Mobility Permit Administrative Regulations". Sanioseca.Gov. last modified 2018. accessed April 28, 2019, http://www.sanioseca.gov/index. aspx?NID=6303.

159. Emily Deruy, "San Jose Imposes New Scooter Regulations", The Mercury News, 2018, accessed April 28, 2019, https://www.mercurynews.com/2018/12/19/san-joseimposes-new-scooter-regulations/.

160. Elizabeth Birriel et al., "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the Transportation Research Board 1773, no. 1 (2001): 47-55. accessed April 10, 2019, https://doi.org/10.3141/1773-06; Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.ith.2017.08.010.

161. "Bay Area Census -- City Of San Jose", Bayareacensus. Ca.Gov. last modified 2010, accessed April 29, 2019, http:// www.bayareacensus.ca.gov/cities/SanJose.htm.

162. "Safety - Bird", Bird, last modified 2019, accessed April 29, 2019, https://www.bird.co/safety.

163. "Lime Electric Scooter Rentals", Li.Me, last modified 2019, accessed April 29, 2019, https://www.li.me/electric-scooter.

164. Sen Lin et al., "Comparison Study On Operating Speeds Of Electric Bicycles And Bicycles", Transportation Research Record: Journal of the Transportation Research Board 2048. no. 1 (2008): 52-59, accessed April 28, 2019, https://doi. org/10.3141/2048-07.

165. "Bay Area Census -- City Of San Jose", Bayareacensus. Ca.Gov. last modified 2010, accessed April 29, 2019, http:// www.bavareacensus.ca.gov/cities/SanJose.htm.

166. Sen Lin et al., "Comparison Study On Operating Speeds Of Electric Bicycles And Bicycles". Transportation Research Record: Journal of the Transportation Research Board 2048. no. 1 (2008); 52-59, accessed April 28, 2019, https://doi. org/10.3141/2048-07.

167. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https://

doi.org/10.1016/j.jth.2017.08.010; Sen Lin et al., "Comparison Study On Operating Speeds Of Electric Bicycles And Bicycles" Transportation Research Record: Journal of the Transportation Research Board 2048, no. 1 (2008): 52-59, accessed April 28, 2019, https://doi.org/10.3141/2048-07,

168. Ibid.

169. Elijah Chiland, "California Removes Helmet Requirement For Electric Scooters". Blog. Curbed: Los Angeles, 2018, accessed April 28, 2019, https://la.curbed. com/2018/9/21/17884220/bird-lime-scooters-ruleshelmets-california.

170. Tarak K. Trivedi et al., "Injuries Associated With Standing Electric Scooter Use", JAMA Network Open 2, no. 1 (2019): e187381, accessed April 10, 2019, http://:10.1001/ iamanetworkopen.2018.7381.

171. "Bike Sharing - Park SJ", Park SJ, accessed April 29, 2019, https://parksj.org/getting-downtown/bike-sharing/.

172. Tarak K. Trivedi et al., "Injuries Associated With Standing Electric Scooter Use", JAMA Network Open 2, no. 1 (2019); e187381, accessed April 10, 2019, http://:10.1001/ jamanetworkopen.2018.7381.

173. Dick de Waard et al., "Mobile Phone Use While Cycling: Incidence And Effects On Behaviour And Safety", Ergonomics 53, no. 1 (2010): 30-42, accessed April 28, 2019, https://doi. org/10.1080/00140130903381180.

174. lbid.

175. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & UseTrailsandEBikes.pdf. Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010.

176. Bicycle And Pedestrian Safety Brochure, ebook (Santa Clara County: Office of Traffic Safety of California, 2019), accessed April 29, 2019, https://www.ots.ca.gov/wp-content/ uploads/sites/67/2018/09/BicvclePedSafetvBrochure.pdf.

177. Elizabeth Suzanne Wolfe et al., "Distracted Biking", Journal of Trauma Nursing 23, no. 2 (2016): 65-70, accessed April 29. 2019. http://10.1097/JTN.000000000000188.

178. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010.

179. Ibid.

180. Ibid.

181. Tarak K. Trivedi et al., "Injuries Associated With Standing Electric Scooter Use", JAMA Network Open 2, no. 1 (2019); e187381. accessed April 10. 2019. http://:10.1001/ jamanetworkopen.2018.7381.

182. Kevin Fang and Susan Handy, "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & 195. Ibid. Health 7 (2017): 288-297, accessed April 10, 2019, https:// doi.org/10.1016/j.jth.2017.08.010.

183. Tarak K. Trivedi et al., "Injuries Associated With Standing Electric Scooter Use", JAMA Network Open 2, no. 1 (2019): e187381, accessed April 10, 2019, http://:10.1001/ jamanetworkopen.2018.7381.

184. "Shared Micro-Mobility Permit Regulations". Sanioseca. Gov, last modified 2019, accessed April 29, 2019, https:// www.sanjoseca.gov/DocumentCenter/View/82493. 185. "City Of San Jose: Speed Limits", Sanjoseca.Gov, last modified 2015, accessed April 28, 2019, http://sanjoseca. gov/DocumentCenter/View/42952.

186. Ibid.

187. Constance Farrell. "5 Things To Know Before You Ride An Electric Scooter", Blog, City Of Santa Monica, 2018, accessed April 29, 2019, https://www.santamonica.gov/ blog/5-things-to-know-before-you-ride-an-electric-scooter.

188. Todd Burley, Multi-Use Trail Pilot Project, ebook (Seattle: City of Seattle: Parks and Recreation, 2018), accessed April 29, 2019, https://www.seattle.gov/Documents/ Departments/SeattleBicycleAdvisoryBoard/presentations/Multi-

189. "Yes, Electric Bikes Are Legal To Ride On Trails", Practicalcycle.Com, accessed April 29, 2019, https://www. practicalcycle.com/yes-electric-bikes-legal-ride-bike-trail/.

190. "E-Scooter Frequently Asked Questions", City Of Oakland, accessed April 29, 2019, https://www.oaklandca.gov/ resources/e-scooter-frequently-asked-questions.

191. Will Kubzansky, "The Secret Life Of Teenage Outlaws", Blog, The Verge, 2018, accessed April 29, 2019, https://www. theverge.com/2018/9/23/17882996/teens-electric-scooterage-requirement-bird-lime.

192. Tarak K. Trivedi et al., "Injuries Associated With Standing Electric Scooter Use", JAMA Network Open 2, no. 1 (2019): e187381, accessed April 10, 2019, http://:10.1001/ iamanetworkopen.2018.7381.

92

193. "City Of San Jose: Speed Limits", Sanjoseca.Gov, last modified 2015, accessed April 28, 2019, http://sanjoseca. gov/Document Center/View/42952.

194. Portland Bureau of Transportation (PBOT), 2018 E-Scooter Findings Report(Portland: Portland Bureau of Transportation (PBOT), 2018), accessed April 28, 2019, https://www.portlandoregon.gov/transportation/article/709719.

196. Elliot Fishman, "Bikeshare: A Review Of Recent Literature". Transport Reviews 36, no. 1 (2015): 92-113, accessed April 10. 2019. http://10.1080/01441647.2015.1033036.

197. Ibid.

198. Ibid.

199. Ibid.

200. lbid.

201. Melissa Etehad and Tracey Lien, "Santa Monica Tells Scooter Companies: You Can Stav - But There Are New Rules", The Los Angeles Times, 2018, accessed April 28, 2019, https://www.latimes.com/business/technology/la-fi-tnsanta-monica-scooter-rules-20180613-story.html.

202. Lizzy Acker, "I Took An Electric Scooter On A 12.5-Mile Journey Through Portland -- Here's What Happened", Blog, The Oregonian: Live, 2018, accessed April 28, 2019, https:// www.oregonlive.com/portland/2018/07/i took an electric scooter on.html.

203. Tarak K. Trivedi et al., "Injuries Associated With Standing Electric Scooter Use", JAMA Network Open 2, no. 1 (2019): e187381, accessed April 10, 2019, http://:10.1001/ iamanetworkopen.2018.7381.

204. Bicycle And Pedestrian Safety Brochure. Ebook. Santa Clara County: Office of Traffic Safety of California, 2019. Accessed April 29, 2019. https://www.ots.ca.gov/wp-content/ uploads/sites/67/2018/09/BicvclePedSafetvBrochure.pdf.

205. Dick de Waard et al., "Mobile Phone Use While Cycling: Incidence And Effects On Behaviour And Safety", Ergonomics 53, no. 1 (2010): 30-42, accessed April 28, 2019, https://doi. ora/10.1080/00140130903381180.

206. Dick de Waard et al., "Mobile Phone Use While Cycling: Incidence And Effects On Behaviour And Safety", Ergonomics 53. no. 1 (2010): 30-42. accessed April 28. 2019. https://doi. org/10.1080/00140130903381180.

207. John Pucher and Ralph Buehler, "Safer Cycling Through Improved Infrastructure", American Journal of Public Health 106, no. 12 (2016): 2089-2091, accessed February 26, 2019, http://10.2105/AJPH.2016.303507.

208. Ibid.

ty 2018–July.pdf.

210. Jeane Barone, "Are Folding Bicycle Helmets Safe?", Blog, Berkeley Wellness, 2017, accessed December 3, 2018, http:// www.berkeleywellness.com/fitness/injury-prevention/article/ are-folding-bicycle-helmets-safe.

org/10.1016/j.trangeo.2017.11.003.

212. lbid.

213. Gulsah Akar, Nicholas Fischer and Mi Namgung, "Bicycling Choice And Gender Case Study: The Ohio State University", International Journal of Sustainable Transportation 7, no. 5 (2013): 347-365, accessed April 29, 2019, https://doi.org/10. 1080/15568318.2012.673694.

214. Rachel Aldred, James Woodcock and Anna Goodman, "Does More Cycling Mean More Diversity In Cycling?", Transport Reviews 36, no. 1 (2015): 28-44, accessed April 29, 2019, https://doi.org/10.1080/01441647.2015.1014451.

209. Populus, The Micro-Mobility Revolution: The Introduction and Adoption of Flectric Scooters in the United States (San Francisco: Populus, 2018), accessed September 9, 2018, http://research.populus.ai/reports/Populus MicroMobili-

211. Gabriele Prati, "Gender Equality And Women's Participation In Transport Cycling", Journal of Transport Geography 66 (2018): 369-375, accessed April 29, 2019, https://doi.

scooter-regulations/

scooter-vendors.

Dill, Jennifer, and Geoffrey Rose. "Electric Bikes And Transportation Policy". Transportation Research Record: Journal of the Transportation Research Board 2314, no. 1 (2012): 1-6. Accessed April 28, 2019, https://doi.org/10.3141/2314-01.

experiment/.

"E-Scooter Frequently Asked Questions". City Of Oakland. Accessed April 29, 2019. https://www.oaklandca.gov/ resources/e-scooter-frequently-asked-questions.

scooter-rules-20180613-story.html.

doi.org/10.1016/j.jth.2017.08.010.

Farrell, Constance. "5 Things To Know Before You Ride An Electric Scooter". Blog. City Of Santa Monica. 2018. Accessed April 29, 2019, https://www.santamonica.gov/blog/5-thingsto-know-before-you-ride-an-electric-scooter.

Federal Highway Administration (FHWA). Characteristics Of Emerging Road And Trail Users And Their Safety, FHWA-HRT-04-104, Federal Highway Administration, 2004, Accessed April 28, 2019. https://www.fhwa.dot.gov/publications/ research/safety/04104/.

Finnis, K. K., and D. Walton, "Field Observations To Determine The Influence Of Population Size, Location And Individual Factors On Pedestrian Walking Speeds". Ergonomics 51, no.

"Article 6. Electric Personal Assistive Mobility Devices [21280 - 21282]". Leginfo.Legislature.Ca.Gov. Last modified 2002.

1177/0361198105193900112.

faces/codes_displayText.

Aultman-Hall, Lisa, and Jeffrey LaMondia. "Evaluating The Safety Of Shared-Use Paths". Transportation Research Record: Journal of the Transportation Research Board 1939, no. 1 (2005): 99-106. Accessed April 28, 2019. https://doi.org/10.

Accessed April 28, 2019. http://leginfo.legislature.ca.gov/

Bank, Chris. "The Rider Experience Of Micro-Mobility Products: Scooters, Bikes & Beyond". Blog. UX Collective, 2018. Accessed April 28, 2019. https://uxdesign.cc/a-usersexperience-with-micro-mobility-5700c5c5104c.

"Bay Area Census -- City Of San Jose". Bayareacensus. Ca.Gov. Last modified 2010. Accessed April 29, 2019. http:// www.bayareacensus.ca.gov/cities/SanJose.htm.

Bicycle And Pedestrian Safety Brochure. Ebook. Santa Clara County: Office of Traffic Safety of California, 2019. Accessed April 29, 2019. https://www.ots.ca.gov/wp-content/uploads/ sites/67/2018/09/BicyclePedSafetyBrochure.pdf.

"Bike Sharing - Park SJ". Park SJ. Accessed April 29, 2019. https://parksi.org/getting-downtown/bike-sharing/ Birriel, Elizabeth, Juan C. Pernia, Jian John Lu, and Theodore A. Petritsch. "Operational Characteristics Of Inline Skaters". Transportation Research Record: Journal of the Transportation Research Board 1773, no. 1 (2001): 47-55, Accessed April 10, 2019. https://doi.org/10.3141/1773-06.

Brinklow, Adam. "Electric Scooter Companies Discount Trips For Students, Low-Income Riders", Blog, Curbed; San Francisco, 2018, Accessed April 28, 2019, https://sf.curbed. com/2018/7/20/17596320/scooters-san-francisco-lime-

Broach, Joseph, Jennifer Dill, and John Gliebe, "Where Do Cyclists Ride? A Route Choice Model Developed With Revealed Preference GPS Data". Transportation Research Part A: Policy and Practice 46, no. 10 (2012): 17301740. Accessed April 28. 2019. https://doi.org/10.1016/j. tra.2012.07.005.

Burley, Todd. Multi-Use Trail Pilot Project. Ebook. Seattle: City of Seattle: Parks and Recreation, 2018, Accessed April 29, 2019. https://www.seattle.gov/Documents/ Departments/SeattleBicycleAdvisoryBoard/presentations/Multi-UseTrailsandFBikes.pdf.

"Bye Bye, Birdie: Beverly Hills Bans Electric Scooters For 6 Months, Pending Regulation". Losangeles.Cbslocal.Com. Last modified 2018. Accessed April 10, 2019. https://losangeles. cbslocal.com/2018/07/24/beverly-hills-bans-bird-limeelectric-scooters/.

Carroll, Rory. "Are Ride-Share Electric Scooters The Future Of Urban Transport?". The Guardian, 2019. Accessed April 10, 2019. https://www.theguardian.com/cities/2018/apr/25/ electric-scooters-urban-transport-bird-santa-monica-uk.

Chiland, Elijah, "California Removes Helmet Requirement For Electric Scooters". Blog. Curbed: Los Angeles, 2018. Accessed April 28, 2019. https://la.curbed.com/2018/9/21/17884220/ bird-lime-scooters-rules-helmets-california. "City Of San Jose: Speed Limits". Sanjoseca.Gov. Last modified 2015. Accessed April 28. 2019. http://sanioseca.gov/ DocumentCenter/View/42952.

Curry, Melanie. "CA Bill No Longer Redefines E-Scooters, But Removes Helmet Requirements For Over-18". Blog. Streetsblog: California, 2018, Accessed April 28, 2019, https:// cal.streetsblog.org/2018/07/03/ca-bill-no-longer-redefinese-scooters-but-removes-helmet-requirements-for-over-18/.

De Waard, Dick, Paul Schepers, Wieke Ormel, and Karel Brookhuis, "Mobile Phone Use While Cyclina: Incidence And Effects On Behaviour And Safety". Ergonomics 53, no. 1 (2010): 30-42. Accessed April 28, 2019. https://doi. ora/10.1080/00140130903381180.

Deruy, Emily. "San Jose Imposes New Scooter Regulations". The Mercury News, 2018. Accessed April 28, 2019. https:// www.mercurynews.com/2018/12/19/san-jose-imposes-new-

SUIDAT DAIXERS, OR TOO FAST AND TOO FURIOUS?

BIBLIOGRAPHY

Abacarion, Robin, "Cities Crack Down On Electric Scooter Scofflaws, While Trying To Untangle The Chaos Of A New Technology". The Los Angeles Times, 2018. Accessed April 28, 2019. http://www.latimes.com/la-me-abcarian-scooters-2018072-story.html.

Acker, Lizzy. "I Took An Electric Scooter On A 12.5-Mile Journey Through Portland -- Here's What Happened". Blog. The Oregonian: Live, 2018. Accessed April 28, 2019. https:// www.oregonlive.com/portland/2018/07/i took an electric scooter on.html.

Aguilar, Jaileane. "Motorized Scooters Restricted". Spartan Daily, 2019. Accessed April 10, 2019. http://www.sisunews. com/spartan daily/news/article 0cc85350-3486-11e9-968c-3f72417da000.html.

Ai, Yi, Zongping Li, and Mi Gan. "A Solution To Measure Traveler's Transfer Tolerance For Walking Mode And Dockless Bike-Sharing Mode". The Journal of Supercomputing (2018). Accessed April 28, 2019. https://doi.org/10.1007/s11227-017-2211-7.

Akar, Gulsah, Nicholas Fischer, and Mi Namgung. "Bicycling Choice And Gender Case Study: The Ohio State University" International Journal of Sustainable Transportation 7. no. 5 (2013): 347-365. Accessed April 29, 2019. https://doi.org/10 .1080/15568318.2012.673694.

Alberts, Brian, Jamie Palumbo, and Eric Pierce, Vehicle 4 Change: Health Implications Of The Capital Bikeshare Program. Washington D.C.: The George Washington University, Master of Public Policy and Public Administration Program, 2013. Accessed April 28, 2019. http://mobilitylab.org/wpcontent/ uploads/2013/05/2013 CapitalBikeshare HealthSurvey Report.pdf.

Aldred, Rachel, James Woodcock, and Anna Goodman. "Does bird-discounts-permits-students. More Cycling Mean More Diversity In Cycling?". Transport Reviews 36, no. 1 (2015): 28-44. Accessed April 29, 2019. https://doi.org/10.1080/01441647.2015.1014451.

"Detail". Dmv.Ca.Gov. Last modified 2018. Accessed April 29, 2019. https://www.dmv.ca.gov/portal/dmv/detail/vr/scooters. Dickey, Meagan R. "Lime Partners With Segway To Build Electric Scooters". Blog. Tech Crunch, 2018. Accessed April 28, 2019. https://techcrunch.com/2018/05/11/in-a-bid-tocorner-supply-bird-locks-in-exclusive-deals-with-the-biggest-

"Dockless Mobility Vehicle Pilot Permit Program Overview". Denvergov.Org. Last modified 2018. Accessed April 10, 2019. https://www.denvergov.org/content/dam/denvergov/ Portals/705/documents/permits/Dockless-Mobility-Pilot-Permit-Program-Overview June2018.pdf.

Ebby, Brianne. "The U.S. Dockless Mobility Experiment". Blog. Eno Transportation Weekly, 2019, Accessed April 10, 2019, https://www.enotrans.org/article/the-u-s-dockless-mobility-

Etehad, Melissa, and Tracey Lien. "Santa Monica Tells Scooter Companies: You Can Stay — But There Are New Rules". The Los Angeles Times, 2018. Accessed April 28, 2019. https:// www.latimes.com/business/technology/la-fi-tn-santa-monica-

Fang, Kevin, and Susan Handy. "Skate And Die? The Safety Performance Of Skateboard Travel: A Look At Injury Data, Fatality Data, And Rider Behavior", Journal of Transport & Health 7 (2017): 288-297. Accessed April 10, 2019. https:// 6 (2008): 827-842. Accessed April 28, 2019. https://doi. org/10.1080/00140130701812147.

Fishman, Elliot, "Bikeshare: A Review Of Recent Literature". Transport Reviews 36, no. 1 (2015); 92–113, Accessed April 10, 2019. http://10.1080/01441647.2015.1033036.

Gagliano, Michael, Buffalo Bike Share Demonstration Project. Report, Buffalo, New York; Shared Mobility Inc., 2015. Accessed April 28, 2019. https://www.nyserda.ny.gov.

Griswold, Alison. "A US City Thinks Electric Scooters Are "Tovs" And Should Be Ridden On The Sidewalk". Blog. Quartz. 2018. Accessed April 28. 2019. https://az.com/1386718/ denver-says-electric-scooters-are-toys-and-should-beridden-on-the-sidewalk/.

Gutman, David, "Helmets May Be Seattle Law, But Many Bike-Share Riders Don'T Wear Them". The Seattle Times. 2017. Accessed April 28, 2019, https://www.seattletimes.com/ seattle-news/transportation/helmets-may-be-seattle-law-butmany-bike-share-riders-dont-wear-them/.

Jose, Ben. "SFMTA Offers Two Permits For One-Year Powered Scooter Pilot", Blog, SFMTA, 2018, Accessed April 10, 2019, https://www.sfmta.com/blog/sfmta-offers-two-permits-oneyear-powered-scooter-pilot.

Kaufman, Sarah M., and Luke Buttenwieser. The State Of Scooter Sharing In United States Cities. New York: Rudin Center Junk Up The Sidewalks". Blog. Wired Magazine, 2018. for Transportation, 2018. Accessed April 10, 2019. https:// wagner.nvu.edu/files/faculty/publications/Rudin ScooterShare scooters-latch-dont-junk-up-sidewalk/. Aug2018 0.pdf.

Keenen, Sean. "Bird Introduces Scooter Delivery Service". Blog. Curbed: Atlanta, 2018. Accessed April 28, 2019. https:// atlanta.curbed.com/2018/10/9/17953718/bird-scooterdelivery-service-last-mile-shareable-vehicles.

Knoblauch, Richard L., Martin T. Pietrucha, and Marsha Nitzburg. "Field Studies Of Pedestrian Walking Speed And Start-Up Time". Transportation Research Record: Journal of the Transportation Research Board 1538, no. 1 (1996): 27-38. Accessed April 28, 2019, https://doi.org/10.1177/03611981 96153800104.

Kubzansky, Will. "The Secret Life Of Teenage Outlaws". Blog. The Verge, 2018, Accessed April 29, 2019, https://www. theverge.com/2018/9/23/17882996/teens-electric-scooterage-requirement-bird-lime.

Kunkle, Fredrick. "Scooters Set To Return To San Francisco" The Washington Post, 2018, Accessed April 28, 2019, https:// www.washingtonpost.com/transportation/2018/10/14/ scooters-set-return-san-francisco/?noredirect=on&utm term=.43ac8deb7bd9.

Landis, Bruce W., Theodore A. Petritsch, Herman F. Huang, and Ann Hong Do. "Characteristics Of Emerging Road And Trail Users And Their Safety". Transportation Research Record: Journal of the Transportation Research Board 1878, no. 1 (2004): 131-139. Accessed April 10. 2019. https://doi. org/10.3141/1878-16.

"Lime Electric Scooter Rentals", Li.Me, Last modified 2019, Accessed April 29, 2019, https://www.li.me/electric-scooter. Lin, Sen, Min He, Yonglu Tan, and Mingwei He. "Comparison Study On Operating Speeds Of Electric Bicycles

Lin, Sen, Min He, Yonglu Tan, and Mingwei He. "Comparison Study On Operating Speeds Of Electric Bicycles And Bicycles" Transportation Research Record: Journal of the Transportation Research Board 2048, no. 1 (2008): 52-59. Accessed April 28, 2019. https://doi.org/10.3141/2048-07.

Madrigal, Alexis C. "San Francisco's Scooter War Is Over". Blog. Citylab, 2018. Accessed April 28, 2019. https://www. citylab.com/transportation/2018/10/san-franciscos-scooterwar-is-over-and-the-scooters-won/573022/.

Marshall, Aarian. "SF Is Bringing Back Electric Scooters With Limits". Blog. WIRED, 2018. Accessed April 28, 2019. https:// www.wired.com/story/san-francisco-santa-monica-e-scooterpermits/

Marshall, Aarian, "Skip Scooters Get A Latch So They Don't Accessed April 28, 2019. https://www.wired.com/story/skip-

Miller, Sheryl, John A. Molino, Jason F. Kennedy, Amanda K. Emo, and Ann Do. "Segway Rider Behavior". Transportation Research Record: Journal of the Transportation Research Board 2073, no. 1 (2008): 125-132. Accessed April 10, 2019. https://doi.org/10.3141/2073-14.

National Association of City Transportation Officials (NACTO). NACTO Policy 2018 Guidelines For The Regulation And Management Of Shared Active Transportation Version 1. New York, 2018. Accessed April 10, 2019. https://nacto. org/wp-content/uploads/2018/07/NACTO-Shared-Active-Transportation-Guidelines.pdf.

Nelson, Laura J. "L.A. Approves Rules For Thousands Of Scooters, With A 15-Mph Speed Limit And Aid For Low-Income Riders". The Los Angeles Times. 2018. Accessed April 28. 2019. https://www.latimes.com/local/lanow/la-me-lnscooter-vote-20180904-story.html.

"Policies + Laws • Peopleforbikes", Peopleforbikes, Accessed April 28, 2019. https://peopleforbikes.org/our-work/e-bikes/ policies-and-laws/.

Populus. The Micro-Mobility Revolution: The Introduction And Adoption Of Electric Scooters In The United. San Francisco: Populus Al, 2018. Accessed April 10, 2019. http://research. populus.ai/reports/ Populus MicroMobility 2018-July.pdf.

Portland Bureau of Transportation (PBOT). 2018 E-Scooter Findings Report. Portland: City of Portland, 2018. Accessed April 28, 2019, https://www.portlandoregon.gov/transportation/ article/709719.

Prati, Gabriele. "Gender Equality And Women's Participation In Transport Cycling". Journal of Transport Geography 66 (2018): 369-375. Accessed April 29, 2019. https://doi.org/10.1016/j. lime-lyft-uber. jtrangeo.2017.11.003.

Robinson, Melia. "A Startup In The West Coast Scooter Sharing Blog. Curbed: San Francisco, 2019. Accessed April Craize Is Already Worth \$1 Billion - Here's What It's Like To Ride A Bird Scooter", Blog, Business Insider, 2018, Accessed April 28, 2019, https://www.businessinsider.com/bird-electric-streets. scooter-review-2018-4.

"Safety - Bird". Bird. Last modified 2019. Accessed April 29, 2019. https://www.bird.co/safety.

San Francisco Municipal Transportation Agency (SFMTA). Pilot Scooter Share Program Permit Staff Report. San Francisco: San Francisco Municipal Transportation Agency (SFMTA), 2018. Accessed April 10, 2019. https://www.sfmta.com/ sites/default/files/reports-and-documents/2018/04/5-1-18 item 11 pilot scooter share program permit.docx .pdf.

"San Jose, CA - Official Website - Micro Mobility Vendor Page - Shared Micro-Mobility Permit Administrative Regulations". Sanjoseca.Gov. Last modified 2018. Accessed April 28, 2019. http://www.sanioseca.gov/index.aspx?NID=6303.

"Shared Micro-Mobility Permit Regulations". Sanjoseca.Gov. Last modified 2019. Accessed April 29, 2019. https://www. sanjoseca.gov/DocumentCenter/View/82493.

Shieber, Jonathan. "Bird Buys More Scooters". Blog. Tech Crunch, 2018. Accessed April 28, 2019. https://techcrunch. com/2018/05/11/in-a-bid-to-corner-supply-bird-locks-inexclusive-deals-with-the-biggest-scooter-vendors.

Sulek, Julia. "Watch Out, Bay Area, The E-Scooter Revolution Is Coming To A Sidewalk Near You". The Mercury News, 2018. Accessed April 10, 2019. https://www.mercurynews. com/2018/04/11/watch-out-bay-area-the-e-scooterrevolution-is-coming-to-a-sidewalk-near-vou/.

Trivedi, Tarak K., Charles Liu, Anna Liza M. Antonio, Natasha Wheaton, Vanessa Kreger, Anna Yap, David Schriger, and Joann G. Elmore. "Injuries Associated With Standing Electric Scooter Use". JAMA Network Open 2, no. 1 (2019): e187381. Accessed April 10, 2019. http://:10.1001/

jamanetworkopen.2018.7381.

Turner, Shawn M., Wiliam L. Eisele, Robert J. Benz, and Douglas J. Holdener, Travel Time Collection Handbook. Washington D.C.: Federal Highway Administration, 1998. Accessed April 29, 2019. https://www.fhwa.dot.gov/ohim/ tvtw/natmec/00020.pdf.

Walker, Alissa. "Bird, Lime, Lyft, Uber Will All Be Allowed To Operate Scooters In Santa Monica". Blog. Curbed: Los Angeles, 2018. Accessed April 28, 2019. https://la.curbed. com/2018/8/30/17800498/scooters-santa-monica-bird-

Walker, Alissa. "Don'T Ban Scooters. Redesign Streets.". 10, 2019. https://www.curbed.com/word-on-thestreet/2018/7/13/17246060/scooters-uber-lvft-bird-lime-

Walker, Alissa. "Everything You Need To Know About Scooters, Bike Share, Dockless Bikes In Los Angeles". Blog. Curbed: Los Angeles, 2018. Accessed April 28, 2019. https://la.curbed. com/2018/7/31/17623336/bird-lime-scooter-bike-applocations-cost-hours.

Weinert, Jonathan X., Chaktan Ma, Xinmiao Yang, and Christopher R. Cherry. "Electric Two-Wheelers In China". Transportation Research Record: Journal of the Transportation Research Board 2038, no. 1 (2007): 62-68. Accessed April 28, 2019. https://doi.org/10.3141/2038-08.

Weismann, Elena. "Enjoy Bird Scooters While You Can, Because They Likely Won't Be Back Until 2019". The Milwaukee Journal Sentinel, 2019. Accessed April 28, 2019. https://www.jsonline.com/story/news/local/ milwaukee/2018/07/20/facing-ban-bird-scooters-maydisappear-milwaukee-until-2019/803014002.

Weissmann, Elena, "Enjoy Bird Scooters While You Can, Because They Likely Won't Be Back Until 2019". Milwaukee Journal Sentinel, 2019. Accessed April 10, 2019. https:// www.jsonline.com/story/news/local/milwaukee/2018/07/20/ facing-ban-bird-scooters-may-disappear-milwaukeeuntil-2019/803014002/

Wolfe, Elizabeth Suzanne, Sandra Strack Arabian, Janis L. Breeze, and Matthew J. Salzler. "Distracted Biking". Journal of Trauma Nursing 23, no. 2 (2016): 65-70. Accessed April 29, 2019. http://10.1097/JTN.000000000000188.

"Yes, Electric Bikes Are Legal To Ride On Trails". Practicalcycle.Com. Accessed April 29, 2019, https://www. practicalcycle.com/yes-electric-bikes-legal-ride-bike-trail/.